

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Barili, Emilia; Bertoli, Paola; Grembi, Veronica; Rattini, Veronica

Working Paper COVID Angels Fighting Daily Demons? Mental Health of Healthcare Workers and Religion

CINCH series, No. 2021/03

Provided in Cooperation with: CINCH – Health Economics Research Center, Universität Duisburg-Essen (UDE)

Suggested Citation: Barili, Emilia; Bertoli, Paola; Grembi, Veronica; Rattini, Veronica (2021) : COVID Angels Fighting Daily Demons? Mental Health of Healthcare Workers and Religion, CINCH series, No. 2021/03, University of Duisburg-Essen, CINCH - Health Economics Research Center, Essen, https://doi.org/10.17185/duepublico/74583

This Version is available at: https://hdl.handle.net/10419/270593

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



#2021/03

Emilia Barili, Paola Bertoli, Veronica Grembi, and Veronica Rattini

COVID Angels Fighting Daily Demons? Mental Health of Healthcare Workers and Religion

Imprint

EDITOR-IN-CHIEF

Martin Karlsson, Essen

MANAGING EDITOR

Katharina Blankart, Essen

EDITORIAL BOARD

Boris Augurzky, Essen Daniel Avdic, Melbourne (AUS) Jeanette Brosig-Koch, Essen Stefan Felder, Basel Annika Herr, Düsseldorf Nadja Kairies-Schwarz, Düsseldorf Hendrik Schmitz, Paderborn Harald Tauchmann, Erlangen-Nürnberg Jürgen Wasem, Essen

CINCH SERIES

CINCH – Health Economics Research Center Weststadttürme, Berliner Platz 6-8 45127 Essen

www.cinch.uni-due.de cinchseries@cinch-essen.de Phone +49 (0) 201 183 - 3679 Fax +49 (0) 201 183 - 3716

All rights reserved. Essen, Germany, 2014

ISSN 2199-8744 (online)

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.



#2021/03

Emilia Barili, Paola Bertoli, Veronica Grembi, and Veronica Rattini

COVID Angels Fighting Daily Demons? Mental Health of Healthcare Workers and Religion









Emilia Barili,^a Paola Bertoli,^b Veronica Grembi,^c and Veronica Rattini^c COVID Angels Fighting Daily Demons? Mental Health of Healthcare Workers and Religion^{*}

Abstract

Relying on a unique survey of more than 15,000 respondents conducted from June to August 2020 in Italy, we show that priming religiosity in healthcare workers decreases the level of self-assessed mental distress experienced during the first wave of the COVID-19. We show that priming religiosity decreases self-assessed mental distress by 9.5%. Consistent with the idea that religiosity serves as a coping mechanism, this effect is stronger for the more impacted categories (e.g., hospital workers) and for respondents facing more stressful situations, such as being reassigned due to the COVID-19 emergency or working in a COVID-19-related specialty (e.g., emergency care), among others. Moreover, higher effects occurs also among physicians who self-classify as religious, while this distinction does not apply for nurses.

Keywords: Healthcare Workers, Mental Health, COVID-19, Coping Mechanisms, Religiosity

JEL classification: 110, N34, Z12

^{*} We thank for their comments on previous draft of this paper Jeanet Bentzen, Lubomir Cingl, Joaquim Vidiella-Martin, Mirjam Reutter and the participants of the research seminar series at the University of Economics and Business (Prague). We are grateful for the support of the physicians and nurses associations that promoted the diffusion of our survey and all the physicians, nurses, biologists, psychologists, obstetricians, and technicians who took time to fill it out. The survey was approved by the Ethics Committee of the University of Verona (Prot. N. 0221872 - 22/06/2020), while the University of Pavia certified compliance with privacy requirements (Prot. N. 61080 - 15/06/2020). All participants gave their informed consent at the beginning of the survey.







^a University of Genoa.

^b University of Verona.

^c University of Milan.

1 Introduction

An increasing number of studies shows the negative impact of the COVID-19 pandemic on mental health regardless of the different institutional contexts and responses to the emergency, as is apparent from evidence from the US (Adams-Prassl et al., 2020; Giuntella et al., 2021) and the UK (Oreffice and Quintana-Domeque, 2020; Proto and Quintana-Domeque, 2021), to name a few. In the light of these results, the experience of some population groups, among which healthcare workers, is expected to be even more dramatic. Consistent with evidence on past disease outbreaks (*e.g.*, severe acute respiratory syndrome—SARS) (Gershon et al., 2016; Lee et al., 2018; Lu et al., 2006; Patel et al., 2018; Phua et al., 2005; Senga et al., 2016; Shah et al., 2020) and with the effects detected in the overall population, by the end of the first wave (spring 2020) of the COVID-19 pandemic, healthcare personnel had experienced a dramatic deterioration of their mental health, showing clear signs of post traumatic stress disorder (PTSD) and burnout (Cabarkapa et al., 2020; Giusti et al., 2020; Muller et al., 2020; Pappa et al., 2020; Preti et al., 2020; Vindegaard and Benros, 2020).

Understanding the role of potential mechanisms (*e.g.*, sports, social support, sharing experiences, meditation) to help cope with the stress triggered by the traumatic situations has important consequences for both the health of specific categories of workers and the general well-being of the population (Bohlken et al., 2020; Cai et al., 2020; Chew et al., 2020; Dong et al., 2020; Labrague and De Los Santos, 2020; Maraqa et al., 2020; Shechter et al., 2020; Xiao et al., 2020). It goes beyond the specific traumatic situation experienced due to the COVID-19 pandemic, possibly applying to other events which are out of the control of individuals. This paper assesses the effectiveness of a specific coping mechanism that can be used to deal with memories of traumatic events: religiosity. The propensity of people to use religiosity to cope with negative events is known as religious coping (Pargament, 2001).¹ Although we are not the first to investigate the role of religiosity as a coping mechanism against mental distress (Koenig, 1998; Shariff et al., 2016), previous studies have failed to estimate a precise magnitude of the role played by religious coping, limiting their discussion to descriptive analyses.

Isolating the effects of religiosity on health presents notable empirical challenges: in some context like the US, religious people are generally wealthier, enjoy higher levels of education, and are in more stable marriages (Bentzen, 2021; Gruber, 2005). All these characteristics are

¹The idea is that people seeking to reduce distress generated by certain events find support in spirituality (Folkman and Lazarus, 1984). This tendency is in turn explained by existential insecurity theory: religious beliefs and practices equip people with rules and habits that are helpful for coping with problems (Immerzeel and Van Tubergen, 2013), moreover religious allegories and parables provide psychological support (Stolz, 2009; Zapata, 2018) and decrease ambiguity (Brandt and Henry, 2012). In the Italian context, researchers found that individuals more exposed to COVID-19 contagion present higher religiosity, which is expressed through individual prayers or attendance to services (Molteni et al., 2020). In addition to these effects, religion can also provide social support through its community dimension (Ellison and George, 1994; Lim and Putnam, 2010), enhance pro-social behaviours (Shariff et al., 2016), reassurance against death anxiety and promotion of the immortality of the soul (Solomon et al., 1991; Vail et al., 2010). Previous research has widely documented a rise in religiosity as a consequence of natural disasters or economic insecurity (Belloc et al., 2016; Bulbulia, 2004; Chen, 2010; Sibley and Bulbulia, 2012; Sinding Bentzen, 2019; Zapata, 2018).

expected to positively affect individual (objective or self-assessed) physical and mental health. To overcome these empirical challenges, we ran an experiment through an online survey that was administered to healthcare workers in Italy between June 15th and August 31st, 2020, while the pressure of the first wave of the pandemic was fading away. The experiment directly manipulates the salience of individual religiosity through implicit priming (Benjamin et al., 2016). Hence, our original survey included two versions of the same 50-item questionnaire, one with a religious priming (i.e., unscrambling statements including "sacred" words) and one with standard neutral questions.² Participants were randomly assigned to one of these two versions. Overall, we collected more than 15,000 questionnaires, with approximately 5,000 respondents among physicians and more than 9,000 among nurses. Based on the answers to the frequency of feelings of distress in the previous months (e.q., feeling frequently or constantly depressedor anxious), we construct our main outcome of interest, Mental distress, which captures the memories of severe mental distress statuses. In addition, we generate an index for the level of *Concerns*, which groups the potential causes of poor self-assessed mental health (e.g., being concerned for your relatives' health or being concerned about the level of stress in the workplace). Since both measures refer to the period from the end of February to June 15th, 2020, our estimated effects refer to coping with past traumatic events, in the spirit of the literature on dealing with PTSD.

We show that priming religiosity decreases self-assessed mental distress by -9.5% in the full sample (out of the mean of mental distress). Among the components of *Mental distress*, the main effect on physicians is feeling depressed and having sleeping problems. In contrast, the effect on nurses arises from all dimensions of mental distress, including fearing that something bad is about to happen and feeling anxious. The baseline results are stronger in the subsample of respondents working in Northern regions, the area most affected by the virus during the first wave. These findings are robust to controlling for the main socioeconomic characteristics of the respondent, the characteristics of her work status (e.g., working in the public or private sector, working in shifts), her own experience with the virus (e.q., testing positive or working with COVID-19 patients), and the experience of her colleagues (e.q., becoming infected ordying from COVID-19). The results are also robust to the use of regional fixed effects. This is relevant since the Italian healthcare system is organized at the regional level (20 regions), and the intensity of the first wave of the pandemic was remarkably different across regions, with Southern regions only marginally affected. The effects on *Mental distress* are also confirmed through an analysis of the impact on *Concerns*, especially on concerns related to *Stress at the workplace*, Personal health, and Relatives health.

Based on the assumption that the priming effects should be stronger when the traumatic experience has a potentially deeper impact, we address the role of religiosity as a coping mechanism in two ways. First, we run the baseline analysis on subcategories of workers who were, on average, under more distress as far as mental health is concerned (*i.e.*, women) and

²In line with previous works (Benjamin et al., 2016), we include "sacred" words not strictly related with a specific religion belief (*i.e.*, spirit, divine, God, sacred, and prophet).

who were more exposed to the first wave (*i.e.*, hospital workers). Gender difference in medically assessed depression is one of the most robust findings in psychiatric epidemiology (Astbury, 2001). It is confirmed both by considering general populations studies conducted worldwide (Piccinelli and Homen, 1997) and by more specific analysis performed in clinical or community samples and differentiating by racial groups (Gater et al., 1998; Kessler et al., To concerns that the observed gap may be driven by artifactual factors (e.q.)1994). measurement errors driven by gender differences in the attitude to report mental distress) or social factors (e.g., gender norms, gender differences in experiences that shape individual future development), other scholars reply that biological factor (*i.e.*, limbic system hyperactivity) predisposes women to experiment anxiety and depressive symptoms (Parker and Brotchie, 2010). They argue that socio-cultural factors act accelerating or reducing the biological response. In our context, this means that women are more likely to report higher mental distress, and as a consequence, the activation of a coping mechanism should have a larger impact on that group. This assumption is confirmed by our results: priming religiosity generates a reduction in reporting extremely poor mental status of -9.9% in the subgroup of women. Likewise, healthcare workers operating in hospitals were extremely exposed to infected patients and the consequences of a lack of organization. Consistently, the estimated effect on these workers implies a reduction of -11.2% in reporting poor mental status.

As a second step, we analyze the differential response based on the type of situation experienced during the first wave, identifying those situations most correlated with high levels of stress. The results show that the effect of priming is stronger in situations regarded as more stressful (e.g., perceived lack of medical personnel, colleagues testing positive or dying from the virus) and, as such, more likely to produce bad consequences for mental health. When we analyze the heterogeneous effects of the stressful situations across categories of workers, we observe a profession-specific (physicians vs. nurses) response to priming.

Finally, we check whether priming religiosity implicitly activates a strictly defined inner religious identity. This is an important step to clarify how the coping mechanism can be activated with different types of individuals (Akerlof and Kranton, 2000). We consider both a more subjective proxy (*i.e.*, self-classification as a religious person) and a less subjective proxy (*i.e.*, prevalence of religious weddings in the province of birth) to identify respondents with different degrees of religious/spiritual identity. We test the results on several subsamples (*i.e.*, the full sample, females, all hospital workers and hospital workers by profession), observing that when the difference between more and less religious respondents is significant, more religious respondents report a stronger effect of the priming (*i.e.*, they present significantly lower levels of mental distress when treated). However, this result holds uniquely in the subsample of physicians, not explaining, in a robust way, variations in the subsample of nurses. In reference to the estimated effects on nurses, we discuss qualitative evidence related to a media campaign naming and portraying them as *COVID Angels* (see Section 2.3 and Appendix C).

Our results contribute to the literature on COVID-19 and the mental health of healthcare

workers and to the discussion of religiosity as a coping mechanism during the COVID-19 pandemic. Since our analysis do not focus on specific intervention related to COVID-19 (*e.g.*, restrictions to mobility) but rather consider it as a stressful situation out of control, our results could be partially generalized to other unexpected traumatic events. Our results are in line with recent publications which show that Google searches on religious topics increased by 50% during the first wave at a global scale (Bentzen, 2020), while descriptive evidence suggests that Italians who were more heavily exposed to COVID-19 during the outbreak or who were religious relied more on religious coping during the first wave (Molteni et al., 2020).³ However, we improve on the existing evidence by priming religiosity to test the channels and magnitudes of its effects on the perception of distressing situations experienced in the recent past.

The paper is organized as follows. Section 2 describes the main features of the Italian healthcare system, the spread of the COVID-19 pandemic in the country and the institutional/governmental response to it. Section 3 provides a description of our survey and of the related religious priming while defining our outcomes of interest. Section 4 illustrates our dataset, and Section 5 presents our econometric specification and the baseline results. Section 6 examines when and for whom religiosity played a stronger role as a coping mechanism, while Section 7 further investigates the relationship between religious priming and individual spirituality. Section 8 concludes.

2 Institutional Background

2.1 The Healthcare System

In Italy, healthcare services are managed at the local level, with 20 regions in charge of providing services to their residents. Although bound to national standards set by the central government, regions enjoy wide discretion in regulating and organizing healthcare delivery within their borders (Ferré et al., 2014). This has resulted in the creation of public hospital networks based on different combinations of hospitals managed by local health authorities (LHAs), independent hospitals (e.g., teaching hospitals) and private accredited institutions (Anessi-Pessina et al., 2004). The gatekeepers are general practitioners (GPs) for the overall population and pediatricians for people younger than 16. Healthcare workers in public hospitals workers can only work in one facility under the so-called exclusivity clause. Healthcare workers also work in nursing homes for elderly or disabled people, in agencies specialized in emergency care, and in local clinics (e.g., in charge of implementing vaccination campaigns, conducting screenings, or offering counselling visits). In addition to physicians and nurses, healthcare personnel include other professional figures, such as dentists, laboratory technicians,

³One-quarter of the U.S. adult population reports having a stronger faith as a consequence of the pandemic (Gecewicz, 2020), while in Italy, there was an increase in praying during the first wave. The data for Italy come from the preview of the survey by (Garelli, 2020), available at http://www.settimananews.it/chiesa/virus-religiosita-degli-italiani/.

obstetricians, and rehabilitation staff. Even though there are regional heterogeneities, ownership of the healthcare facilities is predominantly public, and public hospitals account, on average, for 86% of the overall available beds (83% in Lombardy, which is one of the regions with the largest private-sector presence) (Istat, 2015).

2.2 The First Wave of COVID-19: The Spread of the Virus

The first wave of the pandemic in Italy officially started with the national government's declaration of the state of emergency on January 31st. However, the acquisition of both invasive and non-invasive instruments and ventilation devices and the allocation of protective masks to healthcare personnel only took place approximately a month after the declaration of the state of the emergency; at the time of this declaration, many still considered COVID-19 a non-harmful virus.⁴ By the end of February 2020, the number of reported positive cases increased dramatically, and the Italian National Health Service classified Lombardy and Veneto (two Northern Italian regions) as *COVID-19 red zones*. This classification implied the adoption of strict quarantine measures, restrictions on movements of persons, and the temporary closure of schools, shops, and industrial activities (Sebastiani et al., 2020). By the beginning of March, several other areas in the North of the country were labelled red zones, and starting from March 9th until mid-May, dramatic restrictions on economic activities and mobility of persons were introduced in the entire country.

As a whole, Italy was heavily damaged by the first wave of the COVID-19 pandemic. From the end of February 2020 to mid-June 2020, the total number of assessed cases was 237,290, leading to the deaths of 34,371 people.⁵ As is apparent from Figure 1, the Northern part of the country was the most affected, with remarkable regional variations in the incidence of COVID-19 mortality, as shown in Figure 2.⁶ A more comprehensive view of the hardship of the first wave comes from the comparison of the total mortality during the months of the first wave with the average mortality registered in the same months during the period 2015-2019. As shown in Figure 3, the mortality in January-February 2020 was smaller than that in previous years, while starting from March 2020, there are variations of more than 100% against the levels in the 2015-2019 period.

http://opendatadpc.maps.arcgis.com/apps/opsdashboard/index.html #/b0c68bce2cce478eaac82fe38d4138b1.

⁴Some politicians, such as the mayors of the cities of Milan and Bergamo (both located in Lombardy) and representatives of political parties or businesses associations encouraged the public to adopt a "business as usual" approach.

⁵These figures are taken from the Ministry of Health website:

⁶In exploring the geography-specific intensity of the COVID-19 outbreak, we consider alternative measures of death rates (Istat and Iss, 2020): the COVID-19 mortality rate adjusted for demographic differences between provinces (Figure 2 (a)); the share of COVID-19 deaths over the total number of deaths registered in the relevant period and location (Figure 2 (b)); the percentage variation in the number of deaths registered in 2020 in comparison with the average value registered in the period 2015-2019 (Figure 3 presents the monthly values of this index).

2.3 The First Wave of COVID-19: The Response to the Virus

While the principles and criteria to contain and mitigate the epidemic through case detection, contact tracing, isolation, physical distancing, and mobility restrictions, as well as equipment expansion and staff redeployment, were nationally formalized, each region organized the implementation in its own way and at its own pace (Binkin et al., 2020). As a consequence, regional responses to the first wave of the COVID-19 pandemic differed considerably in terms of investments and timing. For example, clear differences lay in the use of swab testing and in contact-tracing procedures.⁷ Many regions (*e.g.*, Lombardy) followed the national protocol and tested the symptomatic cases only, while other regions (*e.g.*, Veneto) opted for a large-scale procedure.

Since nearly 20% of the hospitalized cases needed two weeks or more of intensive care (ISS, 2020) and 88% required assisted ventilation (Grasselli et al., 2020), enormous efforts were made to increase the number of beds in intensive care units (ICUs), through, for instance, the conversion of hospital wards into ICUs and the creation of temporary hospitals for the intensive care of COVID-19 patients. As shown in Figure 4, at the peak of the first wave of the pandemic, the saturation rate of the ICU beds in the country was approximately 75%, with that of Lombardy and Piedmont being close to 150% (Fanelli et al., 2020).⁸ These transformations meant that a substantial part of the healthcare staff needed to be reassigned from old tasks to new duties overnight and not always with clear guidelines.

During the first wave, healthcare workers faced an unprecedented emergency situation. The numbers of positive cases rapidly depleted local resources, especially in the Northern regions, making it necessary to reassign some of the medical personnel as contact tracers (Barili et al., 2020). The government had to ask retired staff to return to work and to allocate extra funding for the recruitment of 20,000 workers. The work of healthcare professionals was also threatened by shortages of protective supplies (*e.g.*, gloves, medical masks, goggles, and gowns), further increasing their risk of infection, and by continuous changes in the health protocols. Moreover, their higher risk of contagion was overlooked for most of the first wave of the pandemic.⁹ In contrast to the general public, healthcare workers were also excluded from the preventive quarantine measures prescribed after being in contact with a COVID-19-positive patient, and they could stop working only in cases of respiratory symptoms or if they tested positive (Barili

⁷With respect to contact tracing, the national protocol required to notify the related LHA of all the new cases; in the LHAs, explicitly trained nurses and health staff carried out the so-called epidemiological interview. In addition, LHAs had to follow up, isolate and put under surveillance all the recent and close contacts of the interviewed cases.

⁸Contextually, to control for the entry of highly infectious patients into hospitals and reduce the risk of intrahospital contagion, regional and local authorities also activated special emergency numbers and made agreements with the Red Cross and with non-governmental organizations to recruit additional staff and emergency devices. They allowed only urgent cases to directly access hospitals; they organized pre-triage pathways outside hospitals (De Filippo et al., 2020) and created the Special Unit for the Continuity of Care (USCA) to handle the home care of less severe cases. Additionally, resident physicians in other disciplines and generic licensed physicians were called upon to replace or support GPs who had fallen ill or had been quarantined (Barili et al., 2020).

⁹For most of the first wave, the hazard associated with both asymptomatic and pre-symptomatic cases was not yet generally recognized. At the end of the first wave, up to 10% of Italy's confirmed COVID-19 cases were healthcare workers, and between March 11th and May 8th, 178 physicians died of COVID-19 (FNOMCeO, 2020).

et al., 2020).

For all the above reasons, the resilience, extraordinary effort and commitment shown by Italian healthcare workers soon came into the spotlight. Many pictures of nurses with bloodied and bruised faces from their heavy duties or from falling asleep at their desks circulated online (Figure 18), and social media feeds filled with encouraging, heartfelt and grateful messages. During the hardest times of the stay-at-home orders, when healthcare workers were among the very few categories allowed to be outside, newspapers, TV news, and talk shows constantly referred to healthcare personnel as *heroes* and *angels*. Public figures, such as Pope Francis, thanked all the Italian healthcare workers for their heroic service.¹⁰ One image showing a nurse with angel's wings cradling the Italian peninsula in her arms, a *COVID angel*, became symbolic of healthcare workers' dedication during the first wave (Figure 5). The same image was painted over the walls of one of the most heavily affected hospitals in Lombardy as a sign of gratitude toward the extraordinary work done by these workers.¹¹ This unexpected public reaction mainly underscores the sense of community and gratitude, possibly giving meaning to the hardships of this exhausting work. According to a qualitative study on texts and videos posted by nurses from February 23rd to May 3rd, 2020, on professional social media platforms, it appears that "nurses perceived themselves as angels and heroes, receiving gifts, flowers and food during their shifts, and receiving the gratitude of patients thus, with a perception of a sense of recognition for their work. Nurses highlighted how they appreciate this positive feedback in the hope that it might eradicate the negative stigma created by those who before the pandemic committed verbal and physical aggression, especially in emergency departments" (Fontanini et al., 2021).

3 Survey

3.1 Participants and General Procedure

To understand the impact of religious identity on healthcare workers' self-assessed mental health, we ran an online survey between June 15th and August 31st, 2020. Given the end of the main mobility restrictions imposed on June 3rd, this time window can be considered the post-first wave period in Italy. Before launching the survey, we collected the email addresses of potential participants from different sources: the provincial boards of physicians and nurses repositories (108 provinces), hospital websites, and representative associations, some of which advertised and promoted our survey (see Appendix A, Figure A1 and Table A1). Our final contact list included approximately 265,000 email addresses. Each contact received an initial invitation by email followed by 2 reminders (1 and 2 weeks after the first invitation). In each email, we explained that the survey was about the working conditions of healthcare personnel in Italy and

 $^{^{10} \}rm https://www.vaticannews.va/en/pope/news/2020-06/pope-francis-audience-doctors-health-care-priests-covid-pandemic1.html.$

¹¹Positive initiatives directed in particular at nurses (such as the #WEWITHNURSES campaign) have continued even during the second wave. For instance, the company Barilla changed the packaging of one of its products (*i.e.*, its Abbracci ["Hugs"] cookies) to promote a charity campaign to sustain nurses affected by COVID-19 and their families (Figure A2).

that participation was possible using any electronic device (*i.e.*, PC, tablet or smartphone) with an internet connection. Potential participants were also informed that the expected completion time was approximately 15 minutes.

The survey included two versions of the same questionnaire. One version of the survey incorporated religious priming (see Section 3.2), whereas the other version had neutral priming. Participants were randomly assigned to one of these two versions through a random redirect tool compatible with the Google Forms platform, which was used to design the survey.¹² Overall, we included 50 short questions, whose English translations are available in Appendix B.

During the first week of the survey – between June 15th and June 22nd – we ran a pilot with about a 2% share (*i.e.*, 5,000 contacts, randomly selected) of the full sample to check whether the invitation email and the length and structure of the survey were sufficiently comprehensible and effective for collecting valid responses. No particular concerns arose during the pilot; therefore, starting from June 22nd, the survey was emailed to the full sample.¹³ Figure 6 illustrates the timeline of the survey, and Figure 7 illustrates the trend in the completed questionnaires.

The geographical distribution of the survey responses is shown in Figure 8 (a) for the full sample, which includes 5,077 physicians distributed as in Subfigure (b), and 9,069 nurses distributed as in Subfigure (c). Our main focus was on Northern regions, since they were the most affected by the virus. In Figure 9, we report the number of respondents per profession (*i.e.*, physicians vs. nurses) out of the total number of professionals working in both the private and public sectors (Istat, 2015). Overall, the coverage rate in Northern regions was approximately 6.6% for physicians and 4.5% for nurses.¹⁴

In addition to nurses and physicians, other health workers – safety inspectors, controllers, administrative personnel, biologists, and researchers – were also invited to participate in the survey. This was done to capture the impact of COVID-19 on these professionals, who were often reassigned as contact tracers during the first wave of the pandemic, and to more accurately account for the regional disparities in the availability of healthcare personnel.

¹²The choice of the Google Forms platform was made to ensure a user-friendly interface to reduce the impact of formatting on response times and the probability of respondents making mistakes and thereby avoid jeopardizing participant engagement and response validity.

¹³In the empirical analysis, we pooled the data from the pilot with the full-sample observations since no substantial changes were made to the survey after the pilot.

¹⁴The coverage for physicians ranges between a minimum of 2.7% in Friuli-Venezia-Giulia and a maximum of 11.6% in Veneto, while that for nurses ranges between 0.5% in Valle d'Aosta and 7.2% in Trentino-Alto Adige. Considering the entire country, the response rate is 5.7%, meaning an average coverage of approximately 0.2% and 0.9% for physicians and nurses, respectively. This difference in survey responses across Italian macro areas is quite common (Albano et al., 2020; Mazzoleni et al., 2019; Simione and Gnagnarella, 2020). Overall, it is not possible to compare our response rate and coverage with those of previous studies, as our survey potentially targeted all Italian healthcare workers rather than workers in specific hospitals or geographical areas within a country.

3.2 Religious Priming

The efficacy of a priming treatment can be explained in light of self-categorization theory in psychology (James, 2007; Turner, 2010), formalized also in economics (Akerlof and Kranton, 2000): every individual has multiple social identities based on her religiosity, gender, occupation, etc., and at a particular moment, individual behavior could be influenced more by the norms of the salient identity than by the non-salient ones. Therefore, by using priming to make one identity more salient, it is possible to activate identity-salient norms, shedding new light not only on their effects on behavior but also on which norms are associated with the primed identity. For what concern religious identity, the National Institute of Statistics report that in 2019 79.6% of the resident population self-classify as Christian (74.6% Catholic), 15.3% not religious, 5.1% religious but not-Christian (when looking at Italians only, the 82.2% self-classify as Christian (80.1% Catholic), 16.3% not religious, 1.5% religious but not-Christian).

Following the approach implemented and empirically validated by previous scholars, who show that priming significantly increases the salience of participants' religious identity (Benjamin et al., 2016; Shariff and Norenzayan, 2007; Srull and Wyer, 1979), we create two versions of the same questionnaire. The first version makes salient the participants' religious identity through implicit religious priming (the religious priming group), while the second version primes no particular identity with neutral priming (the control group). The priming mechanism consists of a sentence-unscrambling task where subjects have to unscramble 10 five-word sentences by dropping an extra word from each sentence and creating a four-word phrase that makes grammatical sense. To make the priming subtle enough, five of the scrambled sentences contained religiosity-related words (*i.e.*, spirit, divine, God, sacred, and prophet), while the remaining sentences contained only neutral words. The neutral priming treatment instead used only neutral words in all sentences. This task made participants' religious identity salient using religious words since their semantic relatedness sparked participants' mental associations with religiosity. To check for any experimenter demand effect of the priming, at the end of the survey, we also asked participants to speculate on the objectives of the study, and no one reported anything related to their religious identity.

After being exposed to the treatment, respondents were asked questions related to, among others, their mental health status and their main concerns during the COVID-19 pandemic. Comparing the answers provided in the two versions allows us to estimate the marginal effect of religious identity on workers' self-assessed mental health as defined in Section 3. We also asked participants in both the treatment and the control groups to declare their religious beliefs.¹⁵

¹⁵To measure subjects' religious beliefs, we included a direct question similar to the one from the World Values Survey (WVS): "In daily life, which of the following statements describes you better: (a) I am a practising religious person, (b) I am a religious non-practising person, (c) I don't know if I am religious or not, or (d) I am not a religious person".

3.3 Outcomes Definition

We proxy self-assessed mental health referred to the past experience using several measures. The first and most relevant outcome is *Mental distress*, which is a discrete variable ranging from 0 (*i.e.*, no severe mental distress) to 4 (*i.e.*, severe mental distress). This index is the linear sum of the 4 dummies *Depression*, *Anxiety*, *Fear*, and *Sleeping problems*, which are considered the main dimensions of poor mental health.¹⁶ For each of these dimensions, the corresponding dummy tells us if the respondent experienced that specific feeling (*e.g.*, feeling depressed) *very often* or *always* between the beginning of the COVID-19 first wave and June 15th, 2020. In Tables A2 and A3, Appendix A, we provide a more detailed description of each outcome.

Similarly, to measure the overall level of concerns, we rely on a second index, *Concerns*, which is a discrete variable ranging from 0 (*i.e.*, no severe concerns) to 7 (*i.e.*, severe concerns), summing 7 dummies: *Stress in the workplace*, *Personal health*, *Relatives health*, *Couple problems*, *Nobody to talk*, *Financial problems*, and *Family problems*. For each type of concern, a related dummy captures whether that specific situation was a cause of very frequent or constant concern for the respondent from the beginning of the pandemic to June 15th.

These individual types of concern are related to both the working conditions and the daily life of workers and potentially represent the main causes of mental distress. Hence, it is not surprising that our two composite indexes (*i.e.*, *Mental distress* and *Concerns*) are positively and strongly correlated (correlation equal to 0.6). For this reason, our primary interest lies in the study of *Mental distress*, while we use *Concerns* to provide further evidence of the dimensions of the psychological burden suffered by healthcare workers.

As a robustness check, we provide an alternative definition of the main outcomes of interest based on a principal component analysis (PCA). The alternatively defined indexes are highly correlated with the baseline measures (*i.e.*, a correlation of 0.835 for *Mental distress-Mental distress PCA* and of 0.888 for *Concerns-Concerns PCA*) and confirms our findings (see Table A7).

4 Data Analysis

Balance tests are presented in Table 1 for the full sample, while Tables 2 and 3 show similar tests for the samples of physicians and nurses. In addition to the usual questions on gender, nationality, age, marital status, and the presence of children in the household, we insert extra questions to capture other significant determinants of mental well-being, such as housing dimensions and whether the respondent lives alone (Amerio et al., 2020; Husky et al., 2020; Liu et al., 2020). The square footage of the accommodation provides valuable information in

¹⁶Experiencing *Sleeping problems* is a common self-reported symptom among patients diagnosed for PTSD (Spoormaker and Montgomery, 2008). Studies performed in different contexts proved the high degree of co-occurrence of the four dimensions considered (Stein et al., 2018).

at least two ways. On the one hand, it is an indirect measure of wealth that is not necessarily captured by workers' income (which we control for): an individual earning a low salary could still belong to a wealthy family. On the other hand, it is a good proxy for feasible social distancing among cohabitants, which might affect the level of concern and distress within the household. Living alone may also capture different aspects of the pandemic experience: if a worker lives alone, she might be more (psychologically) overwhelmed, but at the same time, she could be less concerned about the well-being of others, since there is no risk of infecting cohabitants. We also collect information on the presence of other healthcare workers in the family of origin. This may be an additional source of concern since workers' relatives, if they are health professionals, are exposed to a higher risk of infection; however, sharing the same profession and challenges might aid workers in coping with distress through this extra source of support.

Table 1 shows that on average, 70% of our respondents work in a hospital, 3% in a teaching hospital, 15% in the private sector, and 59% self-classify as religious (practising or not practising). A comparison of Tables 2 and 3 shows that nurses are more likely to be younger and female than physicians. Consistently, nurses are also less likely than physicians to have children, to be married or cohabiting, and to live in a large accommodation. At the same time, they are more likely to have changed their workplace in the past to work in a hospital (especially in a non-teaching hospital) and in the private sector. Physicians are less likely to be subject to work shifts, but they did more overtime work during the first wave of the COVID-19 pandemic. Overall, no significant differences emerge on any of the observable socioeconomic characteristics between the treatment and the control groups.

Figure 10 reports the distribution of *Mental distress* and *Concerns* across the three categories of professionals in our survey: physicians, nurses, and *Other* (*i.e.*, obstetricians, psychologists, and lab technicians, among others). Overall, nurses report higher levels of both mental distress and concerns than the other two categories. Figures 11 and 12 report the distribution of *Mental distress* and *Concerns* across treatment groups and by profession (i.e., physicians vs. nurses). We find a significant difference in both outcomes between the treatment and the control groups. Healthcare workers who receive religious priming report lower levels of self-assessed distress and concerns than the control group. This difference comes mainly from nurses, while the effect is smaller for physicians. In Figure 13, we instead plot how the treatment effect varies with workers' geographical area, finding no significant difference. For both mental distress and levels of concern, we find stronger effects, to some extent, in the Northern and Central parts of the country. Notice also that the average level of concern is slightly higher in the Southern part of country, although the 95% confidence intervals overlap.¹⁷

¹⁷In Figure A3, we also consider the distribution of *Panic attacks*, a dummy equal to one if the respondent declares she has experienced at least one panic attack and zero otherwise. Although it is apparent that nurses are more likely than physicians to have experienced a panic attack, there are no substantial differences due to priming. This result is important since panic attacks constitute an objective medical condition, and as such, they are not expected to be impacted by ex post exposure to a coping mechanism.

Finally, considering gender and type of workplace, the main effect of religious priming appears to be driven by females and hospital workers (Figure 14), while working in a COVID-19-related specialty (*i.e.*, ICU, anesthesiology, emergency care, cardiology, pulmonary diseases, infectious diseases) drives the main effect on mental distress only, as shown in Figure 15.

5 The Effects of Religious Priming

We estimate the effect of priming religiosity, mainly focusing on *Mental distress*, using the model in Equation 1 – where *Outcomes* captures the different outcomes as described in Tables A2 and A3 for each healthcare worker i working in region r – and the controls listed in Table 4. In addition, we also control for the COVID-19 death rate, which is based on administrative data for the working province p (108 provinces overall) of the worker, as described in Section 2.2.

$$Outcomes_{ir} = \delta Priming_i + \lambda CovidDeath_{ip} + SES'_i\sigma + Providers'_i\gamma +$$
(1)
$$COVID \ Colleagues'_i\beta + COVID \ Own'_i\pi + \tau_r + \epsilon_{ir}$$

The model accounts for both the worker's own exposure to COVID-19 (COVID19 Own) and colleagues' experience with the virus (COVID19 Colleagues). COVID19 Colleagues measures whether the respondent had some colleagues who were infected or died from the virus, while COVID19 Own looks at her personal experience by assessing whether the respondent was exposed to or tested positive for the virus, worked overtime, was reassigned to a different ward/function, or had to directly take care of COVID-19 patients. Providers_i groups details about the employer, such as, among others, the type of healthcare provider (public vs. private, teaching hospitals, etc.), its quality as perceived by the respondent, the opinion of the respondent on system responsiveness to the emergency, and the lack of personnel as perceived by the respondent in her province of work. Finally, we add region of work fixed effects τ_r to deal with all time-invariant geographical differences, such as population characteristics, macroeconomic factors, cultural attitudes, and organization of the healthcare system.

Figure 16 plots the estimated δ on the full sample and by profession. Making the worker's religious identity salient significantly decreases the perception over past mental distress statuses, which drop by -9.5% (-0.08 estimated δ out of 0.84, the mean value of mental distress). When we distinguish between physicians and nurses, the treatment effect on physicians is slightly smaller than that on nurses: the magnitude of the effect on physicians is approximately -7.6% (-0.051/0.67), while the effect on nurses is -11.2% (-0.109/0.97). Table 5 considers the different components of mental distress for the entire sample and only for the Northern regions. The effect of religious priming is stronger on the level of self-assessed depression and sleeping problems

occurred during the first wave. At the average value of *Depression*, the treatment triggers a reduction of -17.8% (*i.e.*, -0.023/0.129) overall, of -15.6% among physicians, and of -19.7% among nurses. The magnitude increases when we consider the Northern regions only, where the figures become -19%, -18%, and -20.8%, respectively. The impact on *Sleeping problems* varies between -8.6% (full sample) and -10.2% (nurses) in the full sample and between -10.4% (full sample) and -13.4% (nurses) in the sample of Northern regions. The reduction in the levels of *Fear* and *Anxiety* observed in the full sample is mainly driven by nurses. This result is in line with the fact that nurses are more likely to be exposed to the disease and to work with COVID-19 patients, which might increase the risk of feeling anxious or the fear that something bad is about to occur (Buselli et al., 2020; Giusti et al., 2020; Magnavita et al., 2020; Riello et al., 2020). These results seem to highlight that religiosity acts reducing the afterwards burden of stressful period, with a stronger effect the hardest the experience.

As expected given the strong correlation between *Mental distress* and *Concerns*, religious priming also negatively affects the overall level of concern, which decreases by -5.5% (-0.101/1.85) in the full sample (Figure 16). However, the effect is statistically significant only among nurses and is equal to -6.6% (0.144/2.19). Among the 7 variables defining *Concerns*, we focus on the three dimensions that record the highest averages: concerns about personal health, relatives' health and the level of stress in the workplace.¹⁸ As shown in Table 6, the greatest effect (a -9.4% reduction) is observed with respect to respondents' personal health. Concerns about relatives' health and on the level of stress in the workplace decrease by -3.7% and -2.5%, respectively, after religious priming. Note also that the overall results are mainly driven by nurses (especially by those working in the Northern regions).¹⁹

6 On Whom and When the Effect Counts the Most

Our baseline results are consistent with evidence from other fields, such as psychology and psychiatry, where it has been shown that religiosity could be an effective means of coping with stress, since it decreases the activation of distress responses in the human brain (Inzlicht and Tullett, 2010; Inzlicht et al., 2011). From a psychological point of view, the idea is that religious beliefs generally provide alternative interpretations of certain events, giving them specific meanings. These meanings, in turn, provide a sense of purpose or an interpretation that enhances hope and motivation among religious people (Koenig and Larson, 2001). This mechanism could be particularly beneficial for mental health when stress is severe or out of control and originates outside the individual (Frankl, 1959; Strawbridge et al., 1998). Consistent with this approach, we further investigate the idea of religiosity as a coping mechanism by focusing on our primary outcome of interest (*i.e.*, *Mental distress*) and considering, on the one hand, the subgroups of workers who are more likely to be

¹⁸Overall, 50.5% of respondents declare that they have been extremely (*i.e.*, very often or always) concerned about their relatives' health during the first wave, while 44% have been extremely concerned about the level of stress in their workplace and 23% about their personal health.

¹⁹The results for the remaining components of *Concerns* are presented in Table A5 in Appendix A.

psychologically overwhelmed by the emergency and, on the other hand, those situations that are correlated with the highest levels of workers' mental distress (Barili et al., 2020). The general intuition is that if religiosity is a way to cope with stress or, at least in our setting, with self-reported measures of mental distress, then the effect should be stronger for the people on whom and in circumstances when the stress factors are higher.²⁰

Hence, we conduct our subgroup analysis on the subsamples of women and of hospital workers. In general, women are more likely to suffer from depression and anxiety than men, while hospital workers were more exposed to the distress of dealing with the emergency. When we estimate Equation 1 on these subsamples, our expectations are confirmed by the coefficients plotted in Figure 17. Priming religiosity triggers a -9.9% (-0.094/0.95) decrease in mental distress among females and a -11.2% decrease among hospital workers (-0.099/0.88). Consistent results are also observed for concerns, for which the effects are stronger among females (-6.5%) and hospital workers (-6.3%) than on the average population (Figure 17). The analysis of the single component of mental distress, shown in Table 7 for both females and hospital workers, confirms that all components are significantly affected by the treatment, and the main driver seems to be concern about the respondent's own work, which decreases by 2.8 percentage points for females and by 2.5 percentage points for hospital workers.

When we move on to the identification of the most stressful situations correlated with higher levels of distress, we focus on 7 cases: reporting a lack of personnel at the province of work level, being reassigned after the COVID-19 outbreak, having at least one colleague who was infected with or died of COVID-19, being personally exposed to the virus or testing positive, and working in a COVID-19-related specialty before the outbreak. Using these dummies, for which we control in the baseline specification, we estimate Equation 2, in which we add the interaction term $Priming_i * D_{i*}$.

$$Outcomes_{ir} = \delta Priming_i + \omega Priming_i * D_i^* + \lambda CovidDeath_{ip} + SES_i^{'}\sigma + Providers_i^{'}\gamma +$$
(2)
$$COVID \ Colleagues_i^{'}\beta + COVID \ Own_i^{'}\pi + \tau_r + \epsilon_{ir}$$

where D_i is the dummy for each dimension. This means that δ represents the effect of the treatment on those not experiencing the additional stress factor and $\delta+\omega$ on those experiencing it, while ω indicates whether the difference between the two groups is statistically significant. For instance, δ captures the effect of not being reassigned due to the COVID-19 emergency, $\delta+\omega$ is the effect on those reassigned, and ω is the difference between the two groups.

As shown in Table 8, our treatment has a stronger effect when the situation was more stressful to start with (*i.e.*, the estimated ω is mostly negative and statistically significant). For instance,

 $^{^{20}}$ As a robustness test, we check whether our main results could be explained by the fact that treated respondents also decide to take greater advantage of leaves (e.g., sick leave, vacations). The results are available in Table A6 and show a basic null effect of the treatment on the likelihood of using leave from work, which could explain a lower incidence of severe mental distress.

we estimate a stronger effect on those reassigned than on those not reassigned, on those working in a COVID-19 specialty, and on those who tested positive for COVID-19. However, the large majority of these channels notably do not explain the experience of physicians but do explain the perception of not having enough healthcare workers in the province where the respondent works. By contrast, being reassigned, having infected colleagues, testing positive for COVID-19, and working in a COVID-19 specialty drive the effects on nurses. Table 9 replicates the same analysis on the subsamples of females and hospital workers. All the channels except for being personally affected by COVID-19 (*i.e.*, being exposed or testing positive) exert the expected effects among females. Overall, the effect on physicians is driven by those perceiving a lack of personnel. In the sample of hospital workers, the experience of being reassigned plays the largest role for both physicians and nurses, consistent with the fact that workers in hospitals were more likely to be reassigned due to the transformation of wards into COVID-19 wards. Finally, we also test for a heterogeneous response based on administrative data on the distribution of ICU beds in 2018, plotting the coefficients in Figure A4.²¹ The intuition behind this check is that the effect of religious priming should be stronger where the number of ICU beds was lower to start with. Even if the estimated effect goes in that direction, the difference between the two samples (i.e., having more vs. fewer ICU beds) is not always statistically significant, probably because we are not able to obtain access to the real distribution of ICU beds in spring 2020.

We find no evidence of different responses to priming based on the severity of the outbreak immediately before the participation date (Figure A5), and the results are not affected by the day of participation in the survey (Figure A6).²²

7 Effective Religious Priming: A Channel

As shown in previous studies, responsiveness to religious cues is related to individual religious beliefs and cultural values (Norenzayan et al., 2013) and to the self-relevancy of priming (Wheeler et al., 2007). Hence, we address the relationship between priming religiosity and classification of the individual as a religious person. Priming should make elements of religiosity and spirituality already present in the respondent more salient. However, it could also be that the spiritual/religious identity of the individual is not exclusively captured by involvement in strictly defined religious activities. For instance, the literature underlines how not only participating in religious services but also attending guided meditation classes have beneficial effects on mental health, lowering anxiety and depression levels through the reinforcement of spiritual identity (Stanley et al., 2011). Hence, we expect priming to have a greater effect on respondents self-classifying as religious, respondents with a stronger

²¹We consider the most updated information available at the time of the analysis (2018). Data on the total number of ICU beds available and their occupancy by COVID-19 patients during the first wave are currently not available. The national agency AGENAS started reporting this information from September 2020 on.

 $^{^{22}}$ The severity of the outbreak is computed as the average growth in the number of COVID-19 cases in the province of work of the respondent during the week before the participation date. Additionally, the results presented in Figure A6 show that priming is not affected by the inclusion of a control for the day of participation in the survey.

spiritual/religious identity or respondents generally feeling themselves to be somehow related to religiosity/spirituality concepts.

We verify this hypothesis by performing a heterogeneity analysis based on self-classification as a religious person (both practising and not practicing)²³ and on the prevalence of religious weddings in the province of birth of the respondent.²⁴ The first measure indicates the respondent's subjective perception of her religious identity, while the second measure underlines the cultural context of her province of origin. We interact the measures with the treatment to estimate any difference in response based on the spiritual/religious identity of the individual as in Equation 2. The results are presented in Table 10 for the full sample, by profession, and for the subgroups of females and hospital workers. When significant, the estimated difference between more and less spiritual/religious types confirms that more spiritual/religious respondents are more responsive to priming, reporting lower levels of mental distress. This is true in particular for physicians, while the distinction is not significant for nurses.

Nurses tend to categorize themselves as religious less frequently than physicians. Given the nature of the treatment, this would predict a lower effect of religiosity among nurses, but this is not the case according to our baseline results. We argue that the effect on nurses is nevertheless consistent with social identity theory due to the media campaign that focused on nurses and exalted them as *COVID angels*. The strongly significant and negative effect estimated for all subgroups of nurses can be reconciled with the effect of priming by this intense exposure, which may have induced in them a feeling of fulfilment and desire towards pro-social behaviours beyond their individual religious identity as commonly defined but that was still made salient by the priming. When remember the sense of gratitude associated to the media campaign, they may be able to deal better with memories of negative events. This result is very relevant when we think about the potential of ad hoc public campaigns to reinforce the effect of coping mechanisms, especially in times of high distress for specific categories of workers.

8 Conclusions

Epidemic outbreaks like the COVID-19 pandemic, which are negative and highly unpredictable events, generate severe emotional distress. Healthcare workers, who are on the front lines in the treatment of patients and the confinement of the infection, are more at risk of feeling fear, anxiety, and exhaustion and of suffering from stress (Preti et al., 2020).

²³Self-classification as a religious person is a subjective measure and could be biased by personal experiences occurring immediately before the survey or by differences in the value assigned to each option. Indeed, different individuals may label their behavior differently according to the standard that they have in mind (*e.g.*, an individual who grows up in a very religious context but is not actually practising that often may classify himself as non-religious, but he may still maintain his religious identity).

²⁴The religious weddings indicator is the percentage of religious weddings out of the total number of weddings held in the province of birth of the healthcare worker in 2018 (last available year). The data are collected by the Italian Institute of Statistics (ISTAT). We define a dummy based on the median value of the distribution to identify *more religious* and *less religious* provinces.

Based on an experimental setting proposed through an online survey directed at Italian healthcare workers, we verify that religiosity is an effective coping strategy to reduce the levels of reported mental stress and the main concerns that most likely induce distress. Although physicians and nurses tend to use different coping mechanisms in emergency situations (Wong et al., 2005), we show that religious coping remains an effective strategy in both groups (Maraqa et al., 2020; Salman et al., 2020; Shechter et al., 2020).²⁵

From a policy perspective, recent studies have highlighted the importance of supporting healthcare workers not only in terms of provision of adequate protective equipment and working environments but also from a psychological point of view. In particular, the WHO proposes a list of practical tools that can be used to support mental health (WHO, 2020). Among these tools, the use of effective coping strategies is considered one of the most important tools that each individual can adopt to preserve her mental well-being. Given the positive results observed for religious coping, the spiritual sensibilities of each individual should always be respected, including through the definition of adequate spaces in the workplace dedicated to prayers and meditation. We additionally observe that community acknowledgement of the relevant role played by healthcare workers in saving lives and protecting patients may have strong beneficial effects on their individual mental status. Honouring care and healthcare workers is indeed a second tool suggested to lower the psychological distress of health workers (WHO, 2020).

Compliance with Ethical Standards

Conflicts of interest: The authors declare that they have no conflicts of interest.

References

- Adams-Prassl, A., T. Boneva, M. Golin, and C. Rauh (2020). The impact of the coronavirus lockdown on mental health: evidence from the us. *HCEO Working Paper* (2020-30).
- Akerlof, G. A. and R. E. Kranton (2000). Economics and identity. The quarterly journal of economics 115(3), 715–753.
- Albano, D., A. Bruno, F. Bruno, M. Calandri, D. Caruso, A. Clemente, P. Coppolino, D. Cozzi, R. De Robertis, F. Gentili, et al. (2020). Impact of coronavirus disease 2019 (covid-19) emergency on italian radiologists: a national survey. *European radiology* 30(12), 6635–6644.
- Amerio, A., A. Brambilla, A. Morganti, A. Aguglia, D. Bianchi, F. Santi, L. Costantini, A. Odone, A. Costanza, C. Signorelli, et al. (2020). Covid-19 lockdown: Housing built environment's effects on mental health. *International journal of environmental research and public health* 17(16), 5973.
- Anessi-Pessina, E., E. Cant, and C. Joninii (2004). Phasing out market mechanisms in the italian national health service. *Public Money and Management* 24(5), 309–316.

 $^{^{25}}$ Doctors mostly use action planning strategies, while nurses resort more to disengagement and distraction activities (Wong et al., 2005).

- Astbury, J. (2001). Gender disparities in mental health. In Mental health. Ministerial Round Tables 2001, 54th World Health Assemble, pp. 73–92.
- Barili, E., P. Bertoli, V. Grembi, and V. Rattini (2020). On the drivers of the well-being and satisfaction of healthcare workers during the covid-19 pandemic. *Mimeo*.
- Belloc, M., F. Drago, and R. Galbiati (2016). Earthquakes, religion, and transition to self-government in italian cities. *The Quarterly Journal of Economics* 131(4), 1875–1926.
- Benjamin, D. J., J. J. Choi, and G. Fisher (2016). Religious identity and economic behavior. Review of Economics and Statistics 98(4), 617–637.
- Bentzen, J. (2020). In crisis, we pray: Religiosity and the covid-19 pandemic. CEPR Discussion Paper No. DP14824.
- Bentzen, J. (2021). Religiosity and Development. Oxford University Press.
- Binkin, N., S. Salmaso, F. Michieletto, and F. Russo (2020). Protecting our health care workers while protecting our communities during the covid-19 pandemic: a comparison of approaches and early outcomes in two italian regions, italy, 2020. medRxiv.
- Bohlken, J., F. Schömig, M. R. Lemke, M. Pumberger, and S. G. Riedel-Heller (2020). Covid-19 pandemic: Stress experience of healthcare workers-a short current review. *Psychiatrische Praxis* 47(4), 190–197.
- Brandt, M. J. and P. Henry (2012). Psychological defensiveness as a mechanism explaining the relationship between low socioeconomic status and religiosity. *International Journal for the Psychology* of Religion 22(4), 321–332.
- Bulbulia, J. (2004). The cognitive and evolutionary psychology of religion. *Biology and philosophy* 19(5), 655–686.
- Buselli, R., M. Corsi, S. Baldanzi, M. Chiumiento, E. Del Lupo, V. Dell'Oste, C. A. Bertelloni, G. Massimetti, L. Dell'Osso, A. Cristaudo, et al. (2020). Professional quality of life and mental health outcomes among health care workers exposed to sars-cov-2 (covid-19). International journal of environmental research and public health 17(17), 6180.
- Cabarkapa, S., S. E. Nadjidai, J. Murgier, and C. H. Ng (2020). The psychological impact of covid-19 and other viral epidemics on frontline healthcare workers and ways to address it: A rapid systematic review. *Brain, behavior, & immunity-health*, 100144.
- Cai, H., B. Tu, J. Ma, L. Chen, L. Fu, Y. Jiang, and Q. Zhuang (2020). Psychological impact and coping strategies of frontline medical staff in hunan between january and march 2020 during the outbreak of coronavirus disease 2019 (covid-19) in hubei, china. *Medical science monitor: international medical* journal of experimental and clinical research 26, e924171–1.
- Chen, D. L. (2010). Club goods and group identity: Evidence from islamic resurgence during the indonesian financial crisis. *Journal of political Economy* 118(2), 300–354.
- Chew, Q. H., F. L.-A. Chia, W. K. Ng, W. C. I. Lee, P. L. L. Tan, C. S. Wong, S. H. Puah, V. G. Shelat, E.-J. D. Seah, C. W. T. Huey, et al. (2020). Perceived stress, stigma, traumatic stress levels and coping responses amongst residents in training across multiple specialties during covid-19 pandemic—a longitudinal study. *International Journal of Environmental Research and Public Health* 17(18), 6572.

- De Filippo, O., F. D'Ascenzo, F. Angelini, P. P. Bocchino, F. Conrotto, A. Saglietto, G. G. Secco, G. Campo, G. Gallone, R. Verardi, et al. (2020). Reduced rate of hospital admissions for acs during covid-19 outbreak in northern italy. *New England Journal of Medicine*.
- Dong, Z.-Q., J. Ma, Y.-N. Hao, X.-L. Shen, F. Liu, Y. Gao, and L. Zhang (2020). The social psychological impact of the covid-19 pandemic on medical staff in china: A cross-sectional study. *European Psychiatry* 63(1).
- Ellison, C. G. and L. K. George (1994). Religious involvement, social ties, and social support in a southeastern community. *Journal for the scientific study of religion*, 46–61.
- Fanelli, S., G. Lanza, A. Francesconi, and A. Zangrandi (2020). Facing the pandemic: The italian experience from health management experts' perspective. The American Review of Public Administration 50(6-7), 753–761.
- Ferré, F., A. G. de Belvis, L. Valerio, S. Longhi, A. Lazzari, G. Fattore, W. Ricciardi, and A. Maresso (2014). Italy: Health system review. *Health Systems in Transition* 16(4), 1–168.
- FNOMCeO (2020). List of doctors who died during the covid-19 epidemic.
- Folkman, S. and R. S. Lazarus (1984). *Stress, appraisal, and coping.* New York: Springer Publishing Company.
- Fontanini, R., E. Visintini, G. Rossettini, D. Caruzzo, J. Longhini, and A. Palese (2021). Italian nurses' experiences during the covid-19 pandemic: a qualitative analysis of internet posts. *International Nursing Review*.
- Frankl, V. (1959). Man's search for meaning. New York: Simon and Schuster.
- Garelli, F. (2020). Virus e religiosità degli italiani. Mimeo.
- Gater, R., T. M., K. A., T. B.G., M. V.G., and O. M.O. (1998). Sex differences in the prevalence and detection of depressive and anxiety disorders in general health care settings. *Archives of General Psychiatry*.
- Gecewicz, C. (2020). Few americans say their house of worship is open, but a quarter say their faith has grown amid pandemic. *Pew Research Center*.
- Gershon, R., L. A. Dernehl, E. Nwankwo, Q. Zhi, and K. Qureshi (2016). Experiences and psychosocial impact of west africa ebola deployment on us health care volunteers. *PLoS currents 8*.
- Giuntella, O., K. Hyde, S. Saccardo, and S. Sadoff (2021). Lifestyle and mental health disruptions during covid-19. Proceedings of the National Academy of Sciences 118(9).
- Giusti, E. M., E. Pedroli, G. E. D'Aniello, C. S. Badiale, G. Pietrabissa, C. Manna, M. S. Badiale, G. Riva, G. Castelnuovo, and E. Molinari (2020). The psychological impact of the covid-19 outbreak on health professionals: a cross-sectional study. *Frontiers in Psychology 11*.
- Grasselli, G., A. Zangrillo, A. Zanella, M. Antonelli, L. Cabrini, A. Castelli, D. Cereda, A. Coluccello, G. Foti, R. Fumagalli, et al. (2020). Baseline characteristics and outcomes of 1591 patients infected with sars-cov-2 admitted to icus of the lombardy region, italy. *Jama 323*(16), 1574–1581.
- Gruber, J. H. (2005). Religious Market Structure, Religious Participation, and Outcomes: Is Religion Good for You? Advances in Economic Analysis & Policy 5(1).

- Husky, M. M., V. Kovess-Masfety, and J. D. Swendsen (2020). Stress and anxiety among university students in france during covid-19 mandatory confinement. *Comprehensive Psychiatry 102*, 152191.
- Immerzeel, T. and F. Van Tubergen (2013). Religion as reassurance? testing the insecurity theory in 26 european countries. *European sociological review* 29(2), 359–372.
- Inzlicht, M. and A. M. Tullett (2010). Reflecting on god: Religious primes can reduce neurophysiological response to errors. *Psychological Science* 21(8), 1184–1190.
- Inzlicht, M., A. M. Tullett, and M. Good (2011). The need to believe: a neuroscience account of religion as a motivated process. *Religion, brain & behavior* 1(3), 192–212.
- ISS (2020). Epidemia covid-19, aggiornamento nazionale: 23 aprile 2020.
- Istat (2015). HEALTH FOR ALL Database.
- Istat and Iss (2020). Impatto dell'epidemia codiv19 sulla mortalitá totale della popolazione residente. periodo gennaio-maggio 2020. *Rapporto ISS ISTAT July*.
- James, W. (2007). The principles of psychology, Volume 1. Cosimo, Inc.
- Kessler, R., K. McGonagle, and S. Z. S. (1994). Lifetime and 12 month prevalence of DSM-111-R psychiatric disorders in the United States. Archives of General Psychiatry.
- Koenig, H. G. (1998). Religious attitudes and practices of hospitalized medically ill older adults. International Journal of Geriatric Psychiatry 13(4), 213–224.
- Koenig, H. G. and D. B. Larson (2001). Religion and mental health: Evidence for an association. International Review of Psychiatry 13(2), 67–78.
- Labrague, L. J. and J. A. A. De Los Santos (2020). Covid-19 anxiety among front-line nurses: Predictive role of organisational support, personal resilience and social support. *Journal of nursing management* 28(7), 1653–1661.
- Lee, S. M., W. S. Kang, A.-R. Cho, T. Kim, and J. K. Park (2018). Psychological impact of the 2015 mers outbreak on hospital workers and quarantined hemodialysis patients. *Comprehensive psychiatry 87*, 123–127.
- Lim, C. and R. D. Putnam (2010). Religion, social networks, and life satisfaction. American sociological review 75(6), 914–933.
- Liu, Y., H. Chen, N. Zhang, X. Wang, Q. Fan, Y. Zhang, L. Huang, B. Hu, and M. Li (2020). Anxiety and depression symptoms of medical staff under covid-19 epidemic in china. *Journal of Affective Disorders* 278, 144–148.
- Lu, Y.-C., B.-C. Shu, Y.-Y. Chang, et al. (2006). The mental health of hospital workers dealing with severe acute respiratory syndrome. *Psychotherapy and psychosomatics* 75(6), 370–375.
- Magnavita, N., G. Tripepi, and R. R. Di Prinzio (2020). Symptoms in health care workers during the covid-19 epidemic. a cross-sectional survey. *International journal of environmental research and public health* 17(14), 5218.

- Maraqa, B., Z. Nazzal, and T. Zink (2020). Palestinian health care workers' stress and stressors during covid-19 pandemic: A cross-sectional study. *Journal of Primary Care & Community Health* 11, 2150132720955026.
- Mazzoleni, G., M. Barbareschi, M. Basciu, D. Fassinato, P. Viani, F. Vittadello, M. Truini, G. De Rosa, and S. Mezzopera (2019). National survey of anatomical pathology centres in italy: the questionnaire. *Pathologica-Journal of the Italian Society of Anatomic Pathology and Diagnostic Cytopathology 111*(1), 4–12.
- Molteni, F., R. Ladini, F. Biolcati, A. M. Chiesi, G. M. Dotti Sani, S. Guglielmi, M. Maraffi, A. Pedrazzani, P. Segatti, and C. Vezzoni (2020). Searching for comfort in religion: insecurity and religious behaviour during the covid-19 pandemic in italy. *European Societies*, 1–17.
- Muller, R. A. E., R. S. Ø. Stensland, R. S. van de Velde, et al. (2020). The mental health impact of the covid-19 pandemic on healthcare workers, and interventions to help them: a rapid systematic review. *Psychiatry Research*, 113441.
- Norenzayan, A., J. Henrich, and E. Slingerland (2013). A synthesis. Cultural evolution: Society, technology, language, and religion 12, 365.
- Oreffice, S. and C. Quintana-Domeque (2020). Gender inequality in covid-19 times: Evidence from uk prolific participants. *IZA Discussion Paper*.
- Pappa, S., V. Ntella, T. Giannakas, V. G. Giannakoulis, E. Papoutsi, and P. Katsaounou (2020). Prevalence of depression, anxiety, and insomnia among healthcare workers during the covid-19 pandemic: A systematic review and meta-analysis. *Brain, behavior, and immunity.*
- Pargament, K. I. (2001). The psychology of religion and coping: Theory, research, practice. Guilford press.
- Parker, G. and H. Brotchie (2010). Gender differences in depression.
- Patel, R. S., R. Bachu, A. Adikey, M. Malik, and M. Shah (2018). Factors related to physician burnout and its consequences: a review. *Behavioral Sciences* 8(11), 98.
- Phua, D., H. Tang, and K. Tham (2005). Coping responses of emergency physicians and nurses to the 2003 severe acute respiratory syndrome outbreak. Academic emergency medicine 12(4), 322–328.
- Piccinelli, M. and F. Homen (1997). Gender differences in the epidemiology of affective disorders and schizophrenia. *Geneva: World Health Organization*.
- Preti, E., V. Di Mattei, G. Perego, F. Ferrari, M. Mazzetti, P. Taranto, R. Di Pierro, F. Madeddu, and R. Calati (2020). The psychological impact of epidemic and pandemic outbreaks on healthcare workers: rapid review of the evidence. *Current psychiatry reports* 22(8), 1–22.
- Proto, E. and C. Quintana-Domeque (2021). Covid-19 and mental health deterioration by ethnicity and gender in the uk. *PloS one 16*(1), e0244419.
- Riello, M., M. Purgato, C. Bove, D. MacTaggart, and E. Rusconi (2020). Prevalence of post-traumatic symptomatology and anxiety among residential nursing and care home workers following the first covid-19 outbreak in northern italy. *Royal Society Open Science* 7(9), 200880.

- Salman, M., M. H. Raza, Z. U. Mustafa, T. M. Khan, N. Asif, H. Tahir, N. Shehzadi, and K. Hussain (2020). The psychological effects of covid-19 on frontline healthcare workers and how they are coping: a web-based, cross-sectional study from pakistan. medRxiv.
- Sebastiani, G., M. Massa, and E. Riboli (2020). Covid-19 epidemic in italy: evolution, projections and impact of government measures. *European journal of epidemiology* 35(4), 341.
- Senga, M., K. Pringle, A. Ramsay, D. M. Brett-Major, R. A. Fowler, I. French, M. Vandi, J. Sellu, C. Pratt, J. Saidu, N. Shindo, and D. G. Bausch (2016). Factors underlying ebola virus infection among health workers, kenema, sierra leone, 2014–2015. *Clinical Infectious Diseases* 63(4), 454–459.
- Shah, K., D. Kamrai, H. Mekala, B. Mann, K. Desai, and R. S. Patel (2020). Focus on mental health during the coronavirus (covid-19) pandemic: applying learnings from the past outbreaks. *Cureus* 12(3).
- Shariff, A. F. and A. Norenzayan (2007). God is watching you: Priming god concepts increases prosocial behavior in an anonymous economic game. *Psychological science* 18(9), 803–809.
- Shariff, A. F., A. K. Willard, T. Andersen, and A. Norenzayan (2016). Religious Priming: A Meta-Analysis With a Focus on Prosociality. *Personality and Social Psychology Review* 20(1), 27–48.
- Shechter, A., F. Diaz, N. Moise, D. E. Anstey, S. Ye, S. Agarwal, J. L. Birk, D. Brodie, D. E. Cannone, B. Chang, et al. (2020). Psychological distress, coping behaviors, and preferences for support among new york healthcare workers during the covid-19 pandemic. *General hospital psychiatry 66*, 1–8.
- Sibley, C. G. and J. Bulbulia (2012). Faith after an earthquake: A longitudinal study of religion and perceived health before and after the 2011 christchurch new zealand earthquake. *PloS one* 7(12), e49648.
- Simione, L. and C. Gnagnarella (2020). Differences between health workers and general population in risk perception, behaviors, and psychological distress related to covid-19 spread in italy. Frontiers in psychology 11, 2166.
- Sinding Bentzen, J. (2019). Acts of god? religiosity and natural disasters across subnational world districts. The Economic Journal 129(622), 2295–2321.
- Solomon, S., J. Greenberg, and T. Pyszczynski (1991). A terror management theory of social behavior: The psychological functions of self-esteem and cultural worldviews. In Advances in experimental social psychology, Volume 24, pp. 93–159. Elsevier.
- Spoormaker, V. I. and P. Montgomery (2008). Disturbed sleep in post-traumatic stress disorder: Secondary symptom or core feature? *Sleep Medicine Reviews* 12(3), 169–184.
- Srull, T. K. and R. S. Wyer (1979). The role of category accessibility in the interpretation of information about persons: Some determinants and implications. *Journal of Personality and Social* psychology 37(10), 1660.
- Stanley, M. A., A. L. Bush, M. E. Camp, J. P. Jameson, L. L. Phillips, C. R. Barber, D. Zeno, J. W. Lomax, and J. A. Cully (2011). Older adults' preferences for religion/spirituality in treatment for anxiety and depression. Aging and Mental Health 15(3), 334–343.
- Stein, L. A., E. Goldmann, A. Zamzam, J. M. Luciano, S. R. Messé, B. L. Cucchiara, S. E. Kasner, and M. T. Mullen (2018). Association between anxiety, depression, and post-traumatic stress disorder and outcomes after ischemic stroke. *Frontiers in Neurology* 9(NOV), 890.

- Stolz, J. (2009). Explaining religiosity: towards a unified theoretical model 1. The British journal of sociology 60(2), 345–376.
- Strawbridge, W. J., S. J. Shema, R. D. Cohen, R. E. Roberts, and G. A. Kaplan (1998). Religiosity buffers effects of some stressors on depression but exacerbates others. *Journals of Gerontology: Social Sciences* 53B(3), 18–126.
- Turner, J. C. (2010). Social categorization and the self-concept: A social cognitive theory of group behavior. Key readings in social psychology. Rediscovering social identity.
- Vail, K. E., Z. K. Rothschild, D. R. Weise, S. Solomon, T. Pyszczynski, and J. Greenberg (2010). A terror management analysis of the psychological functions of religion. *Personality and Social Psychology Review* 14(1), 84–94.
- Vindegaard, N. and M. E. Benros (2020). Covid-19 pandemic and mental health consequences: systematic review of the current evidence. *Brain, Behavior, and Immunity*.
- Wheeler, S. C., K. G. DeMarree, and R. E. Petty (2007). Understanding the role of the self in prime-tobehavior effects: The active-self account. *Personality and social psychology review* 11(3), 234–261.
- WHO (2020). Mental health and psychological considerations during the COVID-19 outbreak. WHO Report 1.
- Wong, T. W., J. K. Yau, C. L. Chan, R. S. Kwong, S. M. Ho, C. C. Lau, F. L. Lau, and C. H. Lit (2005). The psychological impact of severe acute respiratory syndrome outbreak on healthcare workers in emergency departments and how they cope. *European Journal of Emergency Medicine* 12(1), 13–18.
- Xiao, H., Y. Zhang, D. Kong, S. Li, and N. Yang (2020). The effects of social support on sleep quality of medical staff treating patients with coronavirus disease 2019 (covid-19) in january and february 2020 in china. *Medical science monitor: international medical journal of experimental and clinical research 26*, e923549–1.
- Zapata, O. (2018). Turning to god in tough times? human versus material losses from climate disasters in canada. *Economics of disasters and climate change* 2(3), 259–281.

Tables and Figures



Figure 1: COVID-19 Death Rate by Macro Area (Jan-May 2020)

Notes: Average COVID-19 death rate by macro area. The *COVID-19 death rate* is a measure computed by the National Institute of Statistics (ISTAT) together with the Istituto Superiore di Sanità (ISS) on administrative data (Istat and Iss, 2020). The index *death rate*, which refers to the period January-May 2020, represents the mortality rate of COVID-19 standardized according to the demographic characteristics of the resident population in each province (values expressed per 100,000 inhabitants).





Notes: Average COVID-19 death rate and share of COVID-19 deaths by region are measures computed by the National Institute of Statistics (ISTAT) together with the Istituto Superiore di Sanità (ISS) on administrative data (Istat and Iss, 2020). The index *death rate*, which refers to the period January-May 2020, represents the mortality rate of COVID-19 standardized according to the demographic characteristics of the resident population in each province (values expressed per 100,000 inhabitants). The *share of COVID-19 deaths* reports the share of deaths of patients who tested positive for the virus over the total number of deaths in the region and relevant period.



Figure 3: Variations in Death Rates by period-region (2020)

Notes: Variations in *death rates* are measures computed by the National Institute of Statistics (ISTAT) together with the Istituto Superiore di Sanità (ISS) on administrative data (Istat and Iss, 2020). They describe the percentage variation in death rates observed in the relevant month in 2020 in comparison with the corresponding average rate reported in the period 2015-2019.



Figure 4: ICU Occupancy during the First Wave of COVID-19

Notes: The figure presents the share of regional ICU beds used to treat COVID-19 patients (Fanelli

et al., 2020).

Figure 5: COVID Angels



(a) COVID Angels

(b) Wall of one hospital in Lombardy

Notes: Figure 5 presents a health worker with angel's wings cradling the Italian peninsula in her arms (creator: Franco Rivolli). This image became a symbol of healthcare workers' dedication during the first wave of the epidemic in Italy. The same image was painted over the walls of one of the most heavily affected hospitals in Lombardy (Hospital Papa Giovanni XXIII, Bergamo) as a sign of gratitude toward the extraordinary work done by these workers.

Figure 6: Timeline of the Survey

June 15^{th}	June 22^{nd}	Week 2	Week 3	
$\vdash \sim \sim \sim$	~~~~+			
Start Pilot	End Pilot Invitation – Full sample	First Reminder	Second Reminder	





Notes: Each bar is the absolute number of responses to the survey by participation day. *Participation Date* is the day when the survey was completed.

Figure 8: Response Rates by Region



Notes: The response rate is computed as the number of responses received from each region (20) and professional category out of the total number of individuals contacted in that region per category. In line with previous studies (Albano et al., 2020; Mazzoleni et al., 2019; Simione and Gnagnarella, 2020), Northern regions report higher response rates.



Figure 9: Regional Coverage by Professional Category

Notes: Regional coverage is computed as the absolute number of responses received from each region (20) and professional category out of the total number of professionals registered with official boards in that region per category.



Figure 10: Mental Distress and Concerns among Professions

Notes: *Mental distress* is a measure of severe mental distress and takes values between 0 and 4, with 4 representing the highest level of distress. *Concerns* is a measure of severe levels of concern and takes values between 0 and 7, with 7 representing the highest level of concern.

	Neutral Priming	Religious Priming	Difference
			p-value
Children	0.57	0.56	0.37
	(0.50)	(0.50)	
Age	43.54	43.57	0.88
	(12.75)	(12.49)	
Female	0.64	0.65	0.56
	(0.48)	(0.48)	
Italian	0.98	0.99	0.38
	(0.13)	(0.12)	
Married	0.49	0.49	1.00
	(0.50)	(0.50)	
Home sq. meter > 100	0.51	0.50	0.25
	(0.50)	(0.50)	
Good health status	0.95	0.94	0.36
	(0.23)	(0.23)	
Living alone	0.14	0.15	0.11
	(0.35)	(0.36)	
Same workplace	0.27	0.26	0.15
	(0.44)	(0.44)	
Healthcare worker in family	0.34	0.34	0.81
	(0.47)	(0.47)	
Working in hospital	0.70	0.70	0.26
	(0.46)	(0.46)	
Teaching hospital	0.03	0.03	0.30
	(0.16)	(0.16)	
Private	0.15	0.15	0.87
	(0.35)	(0.35)	
Contract with work shifts	0.78	0.78	0.53
	(0.41)	(0.42)	
COVID-19 overtime	0.68	0.69	0.18
	(0.47)	(0.46)	
Religious person	0.59	0.59	0.78
-	(0.49)	(0.49)	

Table 1: Balance Tests: Full Sample

Notes: For an explanation of the variables, see Table A4.

	Neutral Priming	Religious Priming	Difference
			p-value
Children	0.68	0.66	0.13
	(0.47)	(0.47)	
Age	49.34	49.26	0.81
	(12.29)	(11.93)	
Female	0.50	0.50	0.66
	(0.50)	(0.50)	
Italian	0.99	0.99	0.60
	(0.08)	(0.07)	
Married	0.62	0.62	0.82
	(0.49)	(0.49)	
Home sqr. meter > 100	0.68	0.67	0.21
	(0.47)	(0.47)	
Good health status	0.96	0.95	0.21
	(0.19)	(0.21)	
Living alone	0.13	0.15	0.11
	(0.34)	(0.35)	
Same workplace	0.29	0.27	0.09
	(0.45)	(0.44)	
Healthcare worker in family	0.33	0.33	0.61
	(0.47)	(0.47)	
Working in hospital	0.65	0.67	0.10
	(0.48)	(0.47)	
Teaching hospital	0.06	0.07	0.22
	(0.23)	(0.25)	
Private	0.10	0.11	0.18
	(0.30)	(0.31)	
Wage	0.59	0.57	0.12
-	(0.49)	(0.49)	
Contract with work shifts	0.69	0.70	0.75
	(0.46)	(0.46)	
COVID-19 overtime	0.79	0.79	0.96
	(0.41)	(0.41)	
Religious person	0.57	0.58	0.44
- -	(0.50)	(0.49)	

Table 2: Balance Tests: Physicians

Notes: For an explanation of the variables, see Table A4.
	Neutral Priming	Religious Priming	Difference
			p-value
Children	0.50	0.50	0.94
	(0.50)	(0.50)	
Age	40.30	40.38	0.74
	(11.81)	(11.63)	
Female	0.72	0.73	0.24
	(0.45)	(0.44)	
Italian	0.98	0.98	0.47
	(0.15)	(0.14)	
Married	0.42	0.41	0.85
	(0.49)	(0.49)	
Home sqr. meter > 100	0.41	0.40	0.54
	(0.49)	(0.49)	
Good health status	0.94	0.93	0.75
	(0.24)	(0.25)	
Living alone	0.15	0.15	0.42
	(0.36)	(0.36)	
Same workplace	0.26	0.26	0.59
	(0.44)	(0.44)	
Healthcare worker in family	0.35	0.35	0.94
	(0.48)	(0.48)	
Working in hospital	0.72	0.72	0.86
	(0.45)	(0.45)	
Teaching hospital	0.01	0.01	0.83
	(0.08)	(0.08)	
Private	0.17	0.17	0.54
	(0.38)	(0.38)	
Wage	0.03	0.03	0.30
	(0.16)	(0.17)	
Contract with work shifts	0.83	0.82	0.26
	(0.38)	(0.38)	
COVID-19 overtime	0.62	0.64	0.12
	(0.49)	(0.48)	
Religious person	0.59	0.59	0.78
- *	(0.49)	(0.49)	

Table 3: Balance Tests: Nurses

Notes: For an explanation of the variables, see Table A4.



Notes: *Mental distress* is a measure of severe mental distress and takes values between 0 and 4, with 4 representing the highest level of distress.



Figure 12: Descriptives: Concerns by Priming

Notes: *Concerns* is a measure of severe levels of concern and takes values between 0 and 7, with 7 representing the highest level of concern.



Figure 13: Descriptives by Area

Notes: *Mental distress* is a measure of severe mental distress and takes values between 0 and 4, with 4 representing the highest level of distress. *Concerns* is a measure of severe levels of concern and takes values between 0 and 7, with 7 representing the highest level of concern. The areas represent the working area of the respondent.



Figure 14: Descriptives by Gender and Workplace

Notes: *Mental distress* is a measure of severe mental distress and takes values between 0 and 4, with 4 representing the highest level of distress. *Concerns* is a measure of severe levels of concern and takes values between 0 and 7, with 7 representing the highest level of concern.



Figure 15: Descriptives by COVID-19 Specialty

Notes: *Mental distress* is a measure of severe mental distress and takes values between 0 and 4, with 4 representing the highest level of distress. *Concerns* is a measure of severe levels of concern and takes values between 0 and 7, with 7 representing the highest level of concern. COVID - 19 specialty includes ICU, emergency care, anesthesiology, cardiology, pulmonary disease, and infectious disease specialties.

Socioeconomic	Environment perceptions	Network experience	Own experience
Children	Good-quality facility	Infected colleagues	COVID-19 overtime
Age	Lack of personnel	Dead colleagues	Exposed to COVID-19
Female	Prompt response		COVID-19 positive
Italian	Effective response		Taking care of COVID-19 patients
Married			Reassigned due to COVID-19 crisis
Home sq. meter > 100			
Good health status			
Living alone			
Same workplace			
Healthcare worker in family			
Working in hospital			
Teaching hospital			
Private			
Contract with work shifts			
Wage			
Provincial rate of growth of mortality			

Table 4: Controls

Notes: When we refer to the COVID-19 crisis, we mean the first wave that took place in Italy from the end of February 2020 to the beginning of June 2020. *Provincial rate of growth of mortality* represents the mortality rate of COVID-19 standardized according to the demographic characteristics of the resident population in each province (values expressed per 100,000 inhabitants).

Figure 16: **Results**



Notes: The plotted coefficients refer to the estimated value of religious priming as in Equation 1. Controls are those listed in Table 4. Standard errors are clustered at the working region level. Confidence intervals at 95%.



		All Re	gions			Only Northe	rn Regions	
	Depression	Fear	Anxiety	Sleeping problems	Depression	Fear	Anxiety	Sleeping problems
PANEL A: All								
Priming	-0.023***	-0.017***	-0.019***	-0.021***	-0.025***	-0.018**	-0.008	-0.026***
0	(0.004)	(0.005)	(0.006)	(0.004)	(0.005)	(0.006)	(0.005)	(0.004)
Obs	14,746	14,746	14,746	14,746	9,523	9,523	9,523	9,523
Mean	0.129	0.225	0.267	0.243	0.13	0.228	0.272	0.249
PANEL B: Physic	cians							
Priming	-0.015**	-0.013	-0.004	-0.020**	-0.018*	-0.004	0.004	-0.021**
-	(0.006)	(0.009)	(0.009)	(0.008)	(0.008)	(0.008)	(0.01)	(0.008)
Obs	4,829	4,829	4,829	4,829	3,156	3,156	3,156	3,156
Mean	0.096	0.167	0.216	0.188	0.098	0.166	0.22	0.191
PANEL C: Nurse	s							
Priming	-0.029***	-0.020***	-0.031***	-0.028***	-0.031***	-0.029***	-0.018**	-0.038***
0	(0.006)	(0.007)	(0.007)	(0.007)	(0.008)	(0.005)	(0.006)	(0.008)
Obs	8,828	8,828	8,828	8,828	5,581	5,581	5,581	5,581
Mean	0.147	0.257	0.295	0.274	0.149	0.264	0.303	0.283
Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Working region FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

Notes: Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1.

		All Regions			Only Northern Regions			
	Stress in the	Personal	Relatives	Stress in the	Personal	Relatives		
	workplace	\mathbf{health}	\mathbf{health}	workplace	\mathbf{health}	\mathbf{health}		
PANEL A: All								
Priming	-0.011*	-0.022***	-0.019**	-0.014**	-0.018*	-0.019		
	(0.006)	(0.007)	(0.008)	(0.006)	(0.008)	(0.012)		
Obs	14,746	14,746	14,746	9,523	9,523	9,523		
Mean	0.441	0.235	0.505	0.447	0.236	0.507		
PANEL B: Physic	cians							
Priming	-0.016	-0.007	0.000	-0.01 0	0.000	0.000		
0	(0.012)	(0.006)	(0.016)	(0.017)	(0.004)	(0.025)		
Obs	4.829	4.829	4.829	3.156	3.156	3.156		
Mean	0.352	0.16	0.412	0.357	0.168	0.421		
PANEL C: Nurses	S							
Priming	-0.011*	-0.029***	-0.028***	-0.022***	-0.026*	-0.030**		
	(0.006)	(0.009)	(0.009)	(0.005)	(0.013)	(0.009)		
Obs	8,828	8,828	8,828	5,581	5,581	5,581		
Mean	0.508	0.284	0.57	0.52	0.284	0.571		
Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark		
Working region FE	✓ ✓	\checkmark	√	\checkmark	√ √	\checkmark		

Table 6: Priming on Concerns: Main Individual Causes and Geographical Areas

Notes: Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1.

Figure 17: Results for Subgroups



Notes: The plotted coefficients refer to the estimated value of religious priming as in Equation 1. Controls are those listed in Table 4. Standard errors are clustered at the working region level. Confidence intervals at 95%.

Figure 18: Media Campaign



(a) Recruitment of nurses

(b) Nurse at the end of her shift

Notes: Figure 18 (a) was published on the official website of the Ministry of Health (http://www.salute.gov.it/portale/ news/p3_2_1_1_1.jsp?lingua=italiano&menu=notizie&p=dalministero&id=4348). It refers to the recruitment of nurses for the most affected hospitals. Figure 18 (b) reports a picture first published on Twitter that became a symbol of the dedication of nurses. It presents a nurse from the hospital of Cremona, Lombardy, who fell asleep at the end of a shift.

Table 7: Priming on Mental Distress: Individual Causes for Females and Hospital Workers

=

PANEL A: Female	es			
PANEL A.1: Mair	a Causes of Mental Distre	ss		
	Depression	Fear	Anxiety	Sleeping problems
Priming	-0.030***	-0.019***	-0.023**	-0.022**
	(0.004)	(0.005)	(0.010)	(0.008)
Obs	9,635	9,635	9,635	9,635
Mean	0.14	0.25	0.29	0.27
PANEL A.2: Mair	a Causes of Concern			
	Stress in the workplace	Own health	Relatives' health	
Priming	-0.020**	-0.028***	-0.023***	
0	(0.007)	(0.009)	(0.006)	
Obs	9,635	9,635	9,635	
Mean	0.47	0.26	0.55	
PANEL B: Hospita	al Workers			
PANEL B.1: Main	a Causes of Mental Distre	SS		
	Depression	Fear	Anxiety	Sleeping problems
Priming	-0.026***	-0.020**	-0.026***	-0.027***
	(0.003)	(0.007)	(0.008)	(0.004)
Obs	10,306	10,306	10,306	10,306
Mean	0.13	0.23	0.27	0.25
DANDI DA M.	G (G			
PANEL B.2: Main	Causes of Concern			
	Stress in the workplace	Own health	Relatives' health	
Priming	-0.015**	-0.025***	-0.024**	
	(0.006)	(0.007)	(0.010)	
Obs	$10,\!306$	10,306	10,306	
Mean	0.47	0.24	0.53	
Controls		1	./	./
Working region FE	v v	v v	v V	v v
	•	•	•	•

Notes: Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1.

	Lack of personnel	Reassigned	Infected colleagues	Deaths from colleagues	Exposed to COVID-19	Positive for COVID-19	Working in a COVID-19 specialty
PANEL A: All							
Priming	No -0.046^{*} (0.025)	No -0.040** (0.014)	No -0.019 (0.033)	No - 0.071^{***} (0.014)	No -0.084*** (0.011)	No -0.073*** (0.012)	No - 0.056^{***} (0.012)
Priming p interaction	Yes -0.092*** 0.000	Yes -0.231*** 0.000	Yes -0.104*** 0.000	Yes -0.207** 0.024	Yes -0.038 0.532	Yes -0.149*** 0.000	Yes -0.176*** 0.000
Difference	-0.046 (0.031)	-0.191^{***} (0.046)	-0.085^{**} (0.037)	-0.136 (0.094)	$\begin{array}{c} 0.046 \\ (0.061) \end{array}$	-0.076^{**} (0.035)	-0.120^{**} (0.044)
PANEL B: Physic	ians						
Priming	No 0.064 (0.043)	No -0.035 (0.022)	No -0.028 (0.073)	No -0.039 (0.026)	No -0.060*** (0.018)	No - 0.045^{**} (0.016)	No -0.048** (0.021)
Priming p interaction	Yes -0.093*** 0.004	Yes -0.130** 0.015	Yes -0.058*** 0.006	Yes -0.150 0.358	Yes 0.045 0.661	Yes -0.131* 0.082	Yes -0.076 0.416
Difference	-0.157** (0.067)	-0.095 (0.064)	-0.030 (0.086)	-0.111 (0.180)	$\begin{array}{c} 0.105 \\ (0.110) \end{array}$	-0.086 (0.073)	-0.028 (0.104)
PANEL C: Nurses							
Priming	No - 0.118^{**} (0.050)	No -0.051^* (0.027)	No - 0.013 (0.034)	No - 0.099^{***} (0.018)	No - 0.107^{***} (0.017)	No - 0.103^{***} (0.019)	No - 0.070^{***} (0.015)
Priming p interaction	Yes -0.105*** 0.000	Yes -0.286*** 0.000	Yes -0.145*** 0.000	Yes -0.282*** 0.009	Yes -0.120 0.196	Yes -0.147*** 0.001	Yes -0.223*** 0.000
Difference	$\begin{array}{c} 0.013 \\ (0.050) \end{array}$	-0.235*** (0.076)	-0.132*** (0.041)	-0.182 (0.106)	-0.012 (0.092)	-0.044 (0.044)	-0.153^{***} (0.045)
Controls Working region FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

Table 8: Priming on Mental Distress: Potential Drivers

Notes: COVID-19-related specialties include ICU, emergency care, anesthesiology, cardiology, pulmonary disease, and infectious disease specialties. Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1.

	Lack of personnel	Reassigned	Infected colleagues	Deaths from colleagues	Exposed to COVID-19	Positive for COVID-19	Working in a COVID-19 specialty
PANEL A: All Fen	nales						
Priming	No -0.006 (0.034)	No -0.050^{**} (0.019)	No 0.002 (0.037)	No - 0.086^{***} (0.018)	No -0.094*** (0.017)	No -0.089*** (0.021)	No -0.070*** (0.021)
Priming p interaction	Yes -0.125*** 0.000	Yes -0.259*** 0.001	Yes -0.133*** 0.000	Yes -0.230*** 0.004	Yes -0.099 0.280	Yes -0.141*** 0.007	Yes -0.197*** 0.001
Difference	-0.120^{***} (0.039)	-0.209^{***} (0.072)	-0.135^{***} (0.045)	-0.144^{**} (0.067)	-0.004 (0.091)	-0.051 (0.058)	-0.127^{*} (0.062)
PANEL B: Female	Physicians						
Priming	No 0.127 (0.080)	No -0.081* (0.045)	No -0.021 (0.098)	No -0.073 (0.064)	No -0.109^{**} (0.051)	No -0.082 (0.056)	No -0.083 (0.054)
Priming p interaction	Yes -0.167*** 0.009	Yes -0.126 0.251	Yes -0.111* 0.067	Yes -0.246* 0.084	Yes 0.127 0.252	Yes -0.181 0.234	Yes -0.128 0.438
Difference	-0.294^{***} (0.094)	-0.045 (0.085)	-0.089 (0.108)	-0.173 (0.180)	0.237^{**} (0.099)	-0.099 (0.168)	-0.044 (0.173)
PANEL C: Female	Nurses						
Priming	No - 0.083^{**} (0.033)	No -0.082 (0.070)	No -0.041 (0.024)	No 0.006 (0.037)	No - 0.109^{***} (0.019)	No - 0.100^{***} (0.023)	No -0.110^{***} (0.023)
Priming p interaction	Yes -0.212*** 0.001	Yes -0.122*** 0.000	Yes -0.327*** 0.000	Yes -0.157*** 0.000	Yes -0.175 0.161	Yes -0.223** 0.045	Yes -0.130** 0.038
Difference	-0.129 (0.078)	-0.040 (0.080)	-0.286^{***} (0.092)	-0.163^{***} (0.053)	-0.066 (0.119)	-0.123 (0.115)	-0.020 (0.067)
PANEL D: All Hos	spital Worke	ers					
Priming	No -0.048 (0.038)	No - 0.050^{**} (0.022)	No -0.027 (0.051)	No - 0.093^{***} (0.019)	No -0.117*** (0.015)	No -0.089*** (0.020)	No -0.068** (0.025)
Priming p interaction	Yes -0.117*** 0.000	Yes -0.246*** 0.000	Yes -0.122*** 0.000	Yes -0.203* 0.065	Yes 0.072 0.449	Yes -0.193*** 0.000	Yes -0.186*** 0.001
Difference	-0.070^{*} (0.038)	-0.197^{***} (0.061)	-0.096 (0.059)	-0.110 (0.111)	0.189^{*} (0.094)	-0.104^{*} (0.053)	-0.118^{*} (0.063)
PANEL E: Physici	ans Working	g in Hospital	s				
Priming	No 0.125^{**} (0.056)	No -0.050 (0.034)	No 0.022 (0.088)	No -0.061* (0.031)	No - 0.097^{***} (0.025)	No -0.066^{**} (0.027)	No -0.071 (0.042)
Priming p interaction	Yes -0.141*** 0.000	Yes -0.166*** 0.002	Yes -0.101*** 0.002	Yes -0.227 0.244	Yes 0.175 0.283	Yes -0.158 0.223	Yes -0.082 0.372
Difference	-0.266^{***} (0.071)	-0.116^{*} (0.064)	-0.123 (0.095)	-0.166 (0.202)	$\begin{array}{c} 0.272\\ (0.162) \end{array}$	-0.092 (0.132)	-0.011 (0.122)
PANEL F: Nurses	Working in	Hospitals No	No	No	No	No	No
Priming	-0.141^{***} (0.048)	-0.052^{*} (0.030)	-0.053 (0.044)	-0.117^{***} (0.019)	-0.136^{***} (0.019)	-0.113^{***} (0.020)	-0.080*** (0.027)
Priming p interaction	Yes -0.117*** 0.000	Yes -0.299*** 0.001	Yes -0.144*** 0.000	Yes -0.229*** 0.010	Yes -0.006 0.952	Yes -0.200*** 0.000	Yes -0.220*** 0.001
Difference	$\begin{array}{c} 0.024 \\ (0.050) \end{array}$	-0.247** (0.094)	-0.091 (0.058)	-0.111 (0.088)	$0.130 \\ (0.105)$	-0.087 (0.056)	-0.139^{*} (0.074)
Controls Working region FE	√ √	√ √	√ √	√ √	√ √	√ √	√ √

Table 9: Heterogeneities for Females and Hospital Workers

Notes: Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1.

	Entire Sample	Females	Hospital Workers	Entire Sample	Females	Hospital Workers	
	Religious Person		son	Rel	Religious Weddings		
PANEL A: A							
Priming	No -0.089***	No -0.096**	No -0.097***	Less -0.044*	Less -0.052	Less -0.014	
-	(0.029)	(0.042)	(0.032)	(0.023)	(0.038)	(0.028)	
Priming	Yes	Yes	Yes	More 0.104***	More	More 0.152***	
p interaction	0.000	0.012	0.000	0.001	0.000	0.000	
Difference	0.015	0.002	-0.005	-0.060	-0.077	-0.139**	
	(0.038)	(0.068)	(0.037)	(0.047)	(0.057)	(0.053)	
PANEL B: P	hysicians						
Priming	No 0.008	No 0.046	No 0.001	Less 0.030	Less 0.030	Less 0.049	
0	(0.032)	(0.072)	(0.059)	(0.030)	(0.076)	(0.049)	
	Yes	Yes	Yes	More	More	More	
Priming	-0.095***	-0.179**	-0.126***	-0.120***	-0.195***	-0.174***	
p interaction	0.002	0.013	0.002	0.006	0.003	0.004	
Difference	-0.103*	-0.225**	-0.127	-0.150**	-0.225**	-0.223**	
	(0.049)	(0.090)	(0.080)	(0.059)	(0.086)	(0.090)	
PANEL C: N	lurses						
	No	No	No	Less	Less	Less	
Priming	-0.162***	-0.166**	-0.152***	-0.102***	-0.100*	-0.071	
	(0.051)	(0.072)	(0.045)	(0.019)	(0.048)	(0.041)	
	Yes	Yes	Yes	More	More	More	
Priming	-0.071^{***}	-0.077**	-0.103***	-0.155^{***}	-0.228*	-0.151^{***}	
p interaction	0.003	0.050	0.001	0.001	0.058	0.001	
Difference	0.091	0.089	0.049	-0.053	-0.128	-0.080	
	(0.065)	(0.102)	(0.063)	(0.045)	(0.121)	(0.071)	

Table 10: Capturing the Role of Religiosity

Notes: Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1. *Religious Weddings* is the percentage of religious wedding out of the total number of weddings celebrated in the province of birth of the healthcare worker in 2018 (last available year). We define a dummy based on the median value of the distribution to identify *more religious* and *less religious* provinces.

Appendix A

This Appendix provides additional tables and figures, which are also discussed in the paper.

Table A1: Associations participating in the survey

Description	Professionals
Segretariato Italiano Giovani Medici	Physicians
Associazione Anestesisti Rianimatori Ospedalieri Italiani -	Physicians
Emergenza Area Critica	
Organizzazione Sindacale interdisciplinare e apartitica dei	Physicians
Medici Ospedalieri Dipendenti dal S.S.N.	
Associazione Italiana Nursing Sociale	Nurses
Associazione Infermieri di Famiglia e di Comunità	Nurses

Notes: List of associations of healthcare professionals participating in the survey. They provide a valuable contribution sharing the survey through their members.



Figure A1: Some adhesions

Notes: Examples of survey media coverage from associations participating in the study.

Table A2:	Outcomes	Definition -	Mental	Distress
-----------	----------	---------------------	--------	----------

Outcomes	Definition
Mental distress	Continuous from 0 to 4, with 4 the highest value.
	It is the sum of 4 dummies:
	Depression, Fear something bad is about to happen, Anxiety, and Sleeping problems
Depression	Dummy=1 if during the COVID-19 first wave
	you felt depressed or hopeless:
	always or very often and 0 otherwise
Fear	Dummy=1 if during the COVID-19 first wave
	you had been scared that something bad was about to happen:
	always or very often and 0 otherwise
Anxiety	Dummy=1 if during the COVID-19 first wave
	you felt nervous, anxious or edgy:
	always or very often and 0 otherwise
Sleeping problems	Dummy=1 if during the COVID-19 first wave
	you had sleeping problems or problems to get asleep:
	always or very often and 0 otherwise

Notes: When we refer to the COVID-19 first wave, we refer to the first wave which took place in Italy from the end of February 2020 to mid-June 2020.

Outcomes	Definition
Concerns	Continuous from 0 to 7, with 7 the highest value. It is the sum of 7 dummies: Nobody, Family problems, Financial problems, Couple problems, Relatives' health, Personal health, Stress at the workplace
Nobody	Dummy=1 if during the the COVID-19 first wave you were concerned because you did not have anybody to talk to: always or very often and 0 otherwise
Family problems	Dummy=1 if during the COVID-19 first wave you were concerned for family of origin problems: always or very often and 0 otherwise
Financial problems	Dummy=1 if during the COVID-19 first wave you were concerned for financial problems: always or very often and 0 otherwise
Partner problems	Dummy=1 if during the COVID-19 first wave you were concerned for problems with your partner: always or very often and 0 otherwise
Relatives health	Dummy=1 if during the COVID-19 first wave you were concerned for the health of your relatives: always or very often and 0 otherwise
Personal health	Dummy=1 if during the COVID-19 first wave you were concerned for your personal health: always or very often and 0 otherwise
Stress at the workplace	Dummy=1 if during the COVID-19 first wave you were concerned for the level of stress at your workplace: always or very often and 0 otherwise

Table A3: Outcomes Definition - Concerns

 $\it Notes:$ When we refer to the COVID-19 first wave, we refer to the first wave which took place in Italy from the end of February 2020 to mid-June 2020.

Variable	Definition
Children	Dummy=1 if presence of children and 0 otherwise
Age	Age as a continuous dummy
Female	Dummy=1 if female and 0 otherwise
Italian	Dummy=1 if Italian and 0 otherwise
Married	Dummy=1 if married and 0 otherwise
Home sq. meter > 100	Dummy=1 if the home size is more than 100 squared meters and 0 otherwise
Good health status	Dummy=1 if health status is self-classified as good or very good and 0 otherwise
Living alone	Dummy=1 if live alone and 0 otherwise
Same workplace	Dummy=1 if always worked in the same workplace and 0 otherwise
Healthcare worker in family	Dummy=1 if in family of origin there are healthcare workers $(i.e., nurses or physicians)$ and 0 otherwise
Working in hospital	Dummy=1 if work in a hospital and 0 otherwise
Teaching hospital	Dummy=1 if work in a teaching hospital and 0 otherwise
Private	Dummy=1 if work for the private sector and 0 otherwise
Contract with work-shifts	Dummy=1 if have a contract which requires work-shifts and 0 otherwise
Wage	Dummy=1 if the declared wage is above 1,500 (2,500) for nurses (physicians) and 0 otherwise
Provincial COVID19 mortality rate	Provincial COVID19 standardized mortality rate per 100,000 inhabitants
Good quality facility	Dummy=1 if high self-reported quality of the workplace and 0 otherwise
Lack of personnel	Dummy=1 if perception of severe lack of personnel and 0 otherwise
Prompt response	Dummy=1 if perception prompt response from the institution and 0 otherwise
Effective response	Dummy=1 if perception effective response from the institution and 0 otherwise
Infected colleagues	Dummy=1 if colleagues infected from COVID19 and 0 otherwise
Dead colleagues	Dummy=1 if colleagues dead from COVID19 and 0 otherwise
COVID19 overtime	Dummy=1 if did overtime due to COVID19 crisis and 0 otherwise
Exposed to COVID19	Dummy=1 if exposed to COVID19 individuals and 0 otherwise
Positive to COVID19	Dummy=1 if tested positive to COVID19 and 0 otherwise
Taking care of COVID19 patients	Dummy=1 if take care of COVID19 patients and 0 otherwise
Reassigned due to COVID19 crisis	Dummy=1 if reassigned to a different ward due to the COVID19 and 0 otherwise

Table A4: Variables Definition

Notes: When we refer to the COVID19 pandemic, we refer to the first wave which took place in Italy from the end of February 2020 to mid-June 2020. *Wage* in the full sample is a dummy equal to 1 if the wage is above 2,000 euros and 0 otherwise.

		All reg	ions			Only Norther	n Regions	
	Partner relationship	Nobody to talk	Financial troubles	Family troubles	Partner relationship	Nobody to talk	Financial troubles	Family troubles
PANEL A: All								
Priming	-0.021***	-0.008	-0.001	-0.019^{***}	-0.026^{**}	-0.020*	0.001	-0.017**
Obs.	(0.006) 14,746	(0.008) 14,746	(0.007) 14,746	(0.006) 14,746	-0.009) 9,523	(0.008) 9,523	(0.008) 9,523	(0.006) 9,523
Mean	0.186	0.147	0.154	0.186	0.182	0.145	0.139	0.178
PANEL B: Physic	ians							
Priming	0.002	-0.018	0.012	0.000	0.000	-0.021	0.016^{*}	0.008^{*}
	(0.005)	(0.011)	(0.00)	(0.005)	(0.007)	(0.017)	(0.007)	(0.004)
Obs.	4,829	4,829	4,829	4,829	3,156	3,156	3,156	3,156
Mean	0.131	0.102	0.079	0.123	0.133	0.099	0.078	0.121
PANEL C: Nurse								
Priming	-0.037^{***}	-0.003	-0.005	-0.032^{***}	-0.044***	-0.018	-0.008	-0.034***
Obs.	(0.00 <i>1</i>) 8,828	(0.011) 8,828	(0.009) 8,828	(0.009) 8,828	(0.009) 5,581	(0.011) 5,581	5,581	5,581
Mean	0.228	0.179	0.193	0.231	0.222	0.178	0.173	0.222
Controls	>	>	>	>	>	>	>	>
Working region FE	~	~	~	~	~	>	~	>
Notes: Controls are list	ed in Table 4. Sta	undard errors a	are clustered at	t the working re-	gion level. *** $p<0$.01, ** p<0.05	5, * p<0.1.	

4 5 hic ピセワ ÷ • inir Table A5: **Prin**

Figure A2: Packaging dedicated to nurses



Notes: During the second wave, the Barilla company produces a new packaging for the cookies called "Abbracci" -*i.e.*, hugs (Figure A2). The intention is to promote a charity campaign to collect funds to help nurses sick of COVID-19 and their families. The message reported on the back reinforce the concept of gratitude and community directed to nurses that was typical of the first wave. More details on the webpage dedicated: https://www.mulinobianco.it/noicongliinfermieri.



Figure A3: Other outcomes: Panic attack

Notes: *Panic attack* is a dummy equal 1 when the respondent declares to have had at least a panic attack during the COVID-19 first wave.

	All regions	Only North	Females	Hospital workers
PANEL A: All				
Priming	0	0.005	0.005	-0.004
	(0.011)	(0.007)	(0.007)	(0.013)
Obs	14,746	9,523	9,523	$10,\!306$
Mean	0.348	0.366	0.366	0.38
PANEL B: Physic	ians			
Priming	-0.011	-0.01	-0.02	-0.014
1 mining	(0.011)	(0.02)	(0.02)	(0.014)
Obs	4,829	3,156	(0.022) 2,446	3,285
	,	,	,	,
Mean	0.334	0.357	0.369	0.408
PANEL C: Nurses	8			
Priming	0.007	0.012	0.017	-0.003
	(0.016)	(0.008)	(0.012)	(0.018)
Obs	8,828	5,581	$6,\!422$	$6,\!462$
Mean	0.343	0.357	0.349	0.355
Controls	\checkmark	\checkmark	\checkmark	\checkmark
Working region FE	\checkmark	\checkmark	\checkmark	\checkmark

Table A6: Priming on leave permits

Notes: Leave permits are permits for sick leave, vacations, medical conditions of some of the respondent relative (no COVID related). Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p<0.01, ** p<0.05, * p<0.1.



Figure A4: Heterogeneous effects based on the availability of ICU beds

(a) Mental distress & Concerns

(b) Mental distress by profession



(c) Concerns by profession

Notes: Using data from the Ministry of Health, we construct a measure of ICU beds per resident population per province in 2018 (the latest available year), and we define $More \ ICU \ beds$ as those provinces with above the median of the distribution of ICU beds. The Estimations from Equation 1. Confidence intervals at 95%.

Mental distress Concerns Physicians mental distress Nurses mental distress -.25 -.2 -.15 -.1 -.05 0 -.14 -.1 -.06 -.02 .02 80 1 - 18 - 14 - 1 - 06 - 02 02 06 05 .1 -.3 -.25 -.2 -.15 - 05 05 Large COVID19 growth Small COVID19 growth Large COVID19 growth Small COVID19 growth

Figure A5: Heterogeneous effects based on the COVID-19 outbreak

(a) Mental distress & Concerns

(b) Mental distress by profession



(c) Concerns by profession

Notes: Using data from the Ministry of Health, we construct a measure of COVID-19 outbreak in the week before the participation in the survey, and we define *Large COVID-19 outbreak* as those provinces with above the median of the distribution of growth in COVID-19 outbreak in the week before the participation. The Estimations from Equation 1. Confidence intervals at 95%.

Figure A6: Results: control for participation date



Notes: Baseline refers to the estimated value of religious priming as in Equation 1. *Control for participation date* refers to the estimated value of religious priming as in Equation 1 when an additional control for the day of participation is included. For both regressions, baseline controls are those listed in Table 4. Standard errors are clustered at the working region level. Confidence intervals at 95%.

	All regio	ons	Only Northern Regions	
	Mental Distress	Concerns	Mental Distress	Concerns
PANEL A: All				
Priming	-0.106***	-0.107***	-0.116***	-0.100*
0	(0.021)	(0.032)	(0.023)	(0.043)
Obs	13,516	14,746	8,739	9,523
Mean	2.858	4.034	2.897	4.003
PANEL B: Physic	ians			
Priming	-0.075***	-0.050	-0.084**	-0.010
-	(0.021)	(0.036)	(0.025)	(0.016)
Obs	4,600	4,829	3,014	$3,\!156$
Mean	2.570	3.492	2.615	3.512
PANEL C: Nurses	5			
Priming	-0.141***	-0.143***	-0.164***	-0.168*
	(0.028)	(0.037)	(0.021)	(0.056)
Obs	7,914	8,828	4,996	$5,\!581$
Mean	3.094	4.409	3.153	4.371
Controls	\checkmark	\checkmark	\checkmark	\checkmark
Working region FE	\checkmark	\checkmark	\checkmark	\checkmark

Table A7: Alternative definition of outcomes (PCA)

Notes: Dependent variables defined by means of a PCA analysis on the 5 causes of mental distress (Mental distress PCA) and 7 causes of concern (Concerns PCA). Controls are those listed in Table 4. Standard errors are clustered at the working region level. *** p < 0.01, ** p < 0.05, * p < 0.1.

Appendix B: The Survey

INTRODUCTION: We are a group of researchers working at various public universities in Italy (the University of Milan, University of Pavia, and the University of Verona) and we are conducting a survey on the working conditions of the healthcare personnel. This investigation is strictly anonymous and independent from any organization or unit.

You will be asked to answer a set of questions on your demographic characteristics, work habits, and working conditions. On average, the survey will take 13 minutes.

All personal data collected with this survey will be treated as strictly confidential, and will not be made available to any third party. Any data analysis or report based on this survey will protect your anonymity and will be used only for scientific publications.

INSTRUCTIONS: The display may not be compatible with some versions of Mozilla Firefox, the recommended browser is Goggle Chrome. If the survey is carried out by telephone, it is recommended to use the horizontal screen mode of your device. Use the *Next* button to proceed. The answers will be saved at the end of the questionnaire, using the *Submit* button.

QUESTIONNAIRE

- 1. Gender
- 2. Age
- 3. Province of birth
- 4. Province of residence
- 5. In general, how would you say your general health condition is:
 - Excellent
 - . Very good
 - . Good
 - . Not very good
 - . Bad
- 6. Do you suffer from any particular health condition or chronic pathology?
- 7. Do you have an Italian nationality?
- 8. For how many years have you lived in Italy?
- 9. Marital status
- 10. Do you live alone?
- 11. How many children do you have?
- 12. Approximately your home is:
 - . less than 50 squared meters
 - . between 50 and 100 squared meters
 - . between 100 and 150 squared meters
 - over 150 squared meters
 - . I don't know
- 13. Are there in your family of origin (parents, grandparents, uncles) any:
 - Doctors/Dentists
 - . Nurses
 - . Other healthcare professionals
 - . None of the above
- 14. Year of the diploma/degree for the professional qualification

Appendix B p. 1

- 16. Did you ever carry out a continuous period of study/research/work abroad during your training?
 - ${\scriptstyle \bullet}$ Yes, less than 3 months
 - ${\scriptstyle \bullet}$ Yes, between 3 and 6 months
 - ${\scriptstyle \bullet}$ Yes, between 6 months and 1 year
 - . Yes, more than a year
 - . Never
- 17. Profession
- 18. How many specializations do you have?
- 19. Please indicate your primary specialization
- 20. Your current working condition is:
 - . Full-time employment indefinite-time contract
 - . Full-time definite-term contract
 - . Employed part-time indefinite-time contract
 - . Part-time employment definite-time contract
 - . Trainee / PhD student / Resident Doctor
 - **.** SSN Affiliated
 - . Self-employed professional
 - . Freelancer at a public/private facility
 - **.** Looking for a first job
 - Looking for a job
 - Retired
 - Other:
- 21. [For Physicians only]: Which of the following options best describes your occupation:
 - Continuity of care
 - **.** Territorial Medicine
 - . Family Doctor
 - . General Practitioner (affiliated)
 - . Doctor in Nursing Home
 - **.** Doctor in RSA
 - . Doctor in the pharmaceutical industry
 - . Doctor in the Public Administration
 - . Doctor in the Armed Forces
 - . Hospital Doctor, in a private facility
 - . Hospital Doctor, in a private affiliated structure
 - Hospital Doctor, in a public facility
 - Specialist (affiliated)
 - Private practice specialist
 - . Dentist
 - Pediatrician
 - Researcher University
 - Other:

22. [All the rest] Which of the following options best describes your occupation:

- . Home Worker
- Clinic Worker
- . Worker in hospital private
- . Worker in hospital affiliated private

- Worker in hospital public
- . Worker in Nursing Home
- . Worker in Elderly home
- . Worker in the pharmaceutical industry
- **.** Worker in the Armed Forces
- Primary Care Service
- $\hfill \ . Researcher \ University$
- Other:
- 23. Province of work
- 24. How many years have you been working in the facility where you are currently working?
- 25. Approximately how many places of work have you changed since you started working:
- 26. How many hours of work does your contract specify per week?
- 27. Does your contract specify any work shift?
- 28. On average, how many hours do you work per week (including clinical and non-clinical activities)?

28. Since the start of the COVID-19 emergency, on average, how many hours have you worked per week (including clinical and non-clinical activities)?

29. In what department/ward are you currently working?

30. What is the net salary you received last month? Exclude any other monthly payments (thirteenth, fourteenth, etc.) and extra-payments that you do not receive every month (annual productivity bonuses, overdue payments, allowances for missions, unusual overtime, etc.)

31. Within your structure/ward which function do you have:

- **.** Head of the unit
- Coordinator
- Ward Doctor
- Ward worker (including nurse, physiotherapist ...)
- . Resident
- Consultant
- . Director (general, administrative, health, etc.)
- Does not apply (GP, family doctor...)
- Other:
- 32. How much do you trust the managerial skills of the health director?
- 33. How much do you trust the medical expertise of the health director?
- 34. How much do you trust the health guidelines/protocols provided by:
 - The region where you work
 - . The Ministry of Health
- 35. In the Italian public health system, how do you think matter:
 - Merit / competence
 - Politics
- 36. In general, would you say that the facility where you work is of quality:
 - Excellent
 - Very good
 - . Good
 - Not very good.
 - . Bad
 - Not applicable (e.g., GP, pediatrician)
- 37. How much do you agree with the following statements?
 - . If I could start over, I would not be in this profession

- . If I could start over, I would choose a different field of specialization
- My specialization no longer has the appeal it had at the beginning
- . In general, the area in which I specialized has met my expectations
- 38. How satisfied are you:
 - ${\scriptstyle \bullet}$ with the profession?
 - with your current job?
 - with your current income?
 - . with the balance between private and working life?
 - . with the general working environment and the relationships with the colleagues?
 - . with the general working environment and the relationships with the administrative staff?
 - with the number of hours worked?
 - with the career prospects?
- 39. How exposed do you feel to the following risk factors?
 - . Time pressure or excessive workload
 - Physical threats or violence
 - Bullying or harassment
 - . Poor communication or collaboration within the organization
 - . Difficult-to-treat patients, trainees, and students
 - . Lack of autonomy or control over time or processes
 - Workplace insecurity
 - Professional responsibility

40. In your opinion, is there a lack of medical personnel that could endanger residents' access to care in the province where you work?

41. From the beginning of the COVID-19 emergency until 15/06, how often did you experience the following conditions:

- . Feeling depressed or hopeless
- . Feeling nervous, anxious, or on edge
- Having troubles in sleeping or falling asleep
- . Being a fraid that something bad might happen

42. From the beginning of the COVID-19 emergency until 15/06, have you ever been worried about:

- . Your health
- . The health of your relatives and friends
- . The relationship with your partner (if in a couple)
- . The relationship with your family of origin
- . The stress at your workplace
- **.** Financial problems
- . Not having anyone to talk to

43.From the beginning of the COVID-19 emergency until 15/06, did you have any panic attack (sudden feeling of fear or panic?):

44. From the beginning of the COVID-19 emergency until 15/06, were you:

- **.** Hired for the emergency
- . Fired / Laid off
- . Assigned to a ward/hospital other than your habitual ward/hospital
- . Directly involved in the treatment of COVID-19 patients
- . In contact with COVID-19 positive patients
- . Directly involved in the care of post-COVID-19 patients (no longer positive)
- . In solitary confinement because of the suspicion of a COVID-19 infection

- . In quarantine because infected (positive swab) by COVID-19
- . In quarantine because symptomatic of COVID-19
- . Hospitalized for COVID-19
- . Hospitalized for other reasons from COVID-19 but related to your professional activity

45. Have some of your colleagues been infected and/or hospitalized, or lost their lives due to COVID-19?

 ${\ensuremath{\scriptstyle \bullet}}$ No

- Yes, infected in quarantine
- Yes, infected and hospitalized
- Yes, infected and deceased

46. From the beginning of the COVID-19 emergency until 15/06, have you taken some days off (sickness, holidays, law 104, parental leave)?

. No

- Yes, 1-5 days
- Yes, 6-15 days
- Yes, 15-30 days
- Yes, more than 30 days

47. How do you evaluate the COVID-19 protocols adopted by your facility (province if you do not belong to any facility) in terms of [very positive, somehow positive, average, not very positive, not positive at all] :

- Timing
- ${\scriptstyle \bullet}$ Effectiveness

48. In your daily life, which of the following statements describes you better:

- . I am practicing religious
- . I am non-practicing religious
- . I can't tell if I'm religious or not.
- . I'm not a religious person
- 49. [Optional question] In your opinion, what was the purpose of this survey?
- 50. [Optional question] Do you belong to one of the following groups?
 - **.** Trade union association
 - Scientific association
 - . Association concerning your professional specialization
 - . I do not belong to any group related to my profession
 - Other

A Sentence-Unscrambling Task Instructions

Now we ask you to participate in a simple activity. Starting from a group of 5 words, you have to form a sentence of 4 words, excluding, therefore, one of the initial 5 words. For example: high winds the flies plane \rightarrow the plane flies high. Answers containing no characters or punctuation will not be accepted. Any word order will be accepted, therefore there is no right or wrong answers. Please complete each scramble within 60 seconds.

Sentences - Religious priming

- 1. felt she eradicate spirit the
- 2. dessert divine was fork the
- 3. appreciated presence was imagine her
- 4. more paper it once do
- 5. send I over it mailed
- 6. evil thanks give God to
- 7. yesterday it finished track he
- 8. sacred was book refer the
- 9. reveal the future simple prophets
- 10. prepared somewhat I was retired

Sentences - Neutral priming

- 1. fall was worried she always
- 2. shoes give replace old the
- 3. retrace good have holiday a
- 4. more paper it once do
- 5. send I over it mailed
- 6. saw hammer he the train
- 7. yesterday it finished track he
- 8. sky the seamless blue is
- 9. predictable he shoes his tied
- 10. prepared somewhat I was retired

Appendix C: Mass Media Campaign

In this Appendix, we collected qualitative evidence on the mass media campaign which identifies healthcare personnel to *COVID Angels*. There are mainly titles from newspapers articles spanning from February to June 2020. The title of the article is translated in the title of the figure.

Figure C1: "Coronavirus, the angels of the red zone"



Notes: Article published on Corriere della Sera on February 28, 2020.



Figure C2: "Hospitals, nurses, pharmacies: the angels of the virus"

Notes: Article published on Il Fatto Quotidiano on March 09, 2020.

Figure C3: "The angels of the ambulances leave "On mission to the North""



Notes: Article published on La Repubblica on March 25, 2020.

Figure C4: "The angels of the ambulances, from Rome to the front line"



Notes: Article published on La Repubblica on March 25, 2020.

Figure C5: "Coronavirus, "Heroes Angels": 10 thousands thanks in 24 hours to Areu for doctors and rescuers"



Notes: A reu is the agency for the emergency of the Lombardy region. Article published on $\it Il~Giorno$ on March 27, 2020.



Figure C6: "The angels of intensive care, I was reborn thanks to their smiles"

«Gli Angeli della terapia intensiva sono rinato grazie ai loro sorrisi»

Notes: Article published on L'Eco di Bergamo on April 19, 2020. The province of Bergamo has been among the most affected in the country by the first wave of the pandemic.



Figure C7: "Gloves, gowns, socks. The white angels of the dark"

Notes: Article published on La Nazione on April 22, 2020.

Figure C8: "Behind the "diving suit", the great heart of the Angels in the wards"

IL RINGRAZIAMENTO

Dietro lo "scafandro", il grande cuore degli Angeli in corsia

Gli eroi della Clinica Città Studi: non solo medici e infermieri, anche personale ausiliario e fisioterapisti. Sorrisi, impegno e tanta

passione

Pubblicato il 8 maggio 2020





Gli Angeli della corsia «in trincea» nella Clinica Città Studi di Milano

Notes: Article published on Il Giorno on May 08, 2020.

Figure C9: "International nurses day: this is how Bergamo celebrates the angels of the COVID emergency"



Notes: Article published on Corriere della Sera on May 12, 2020.

Figure C10: "The voice of the sick: thanks to the angels of COVID"



La voce dei malati grazie agli angeli del Covid

Dall'Hospice di Bià ai reparti del Fornaroli: il racconto della dottoressa impegnata a inventarsi le cure palliative per il coronavirus

di FRACESCO PELLEGATTA

Notes: Article published on Il Giorno on May 27, 2020.

Figure C11: "Healthcare workers rebel "We are underpaid angels""

la Repubblica Archivio							
Home	Pubblico	Economia&Finanza	Sport	Spettacoli	Cultura	Мо	
Sei in: Archivio > la Repubblica.it > 2020 > 06 > 09 > I sanitari si ribellano							
I sanitari si ribellano "Siamo							

Notes: Article published on La Repubblica on June 09, 2020.

angeli sottopagati"

Figure C12: "The protest of the nurses, sit-in of the angels without rewards"

IResto del Carlino PESARO CRONACA SPORT COSA FARE EDIZIONI COLORI REGIONI EMILIA ROMAGNA MARCHE COVID DELITTO FAENZA Q HOME PESARO CRONACA LA PROTESTA DEGLI INFERMIERI,... Pubblicato il 27 maggio 2020 La protesta degli infermieri, sit-in degli angeli senza premi Doppio presidio sia a Pesaro che a Fano. Ospedale e Area Vasta: "A giugno le indennità di marzo e aprile". Ma non basta

Notes: Article published on Il Resto del Carlino on June 09, 2020.

di SIMONA SPAGNOLI

Figure C13: "The angels of the pandemic who were and remain precarious"





CINCH working paper series

- Halla, Martin and Martina Zweimüller. Parental Responses to Early Human Capital Shocks: Evidence from the Chernobyl Accident. CINCH 2014.
- 2 Aparicio, Ainhoa and Libertad González. **Newborn Health and the Business Cycle:** Is it Good to be born in Bad Times? CINCH 2014.
- **3** Robinson, Joshua J. **Sound Body, Sound Mind?:** Asymmetric and Symmetric Fetal Growth Restriction and Human Capital Development. CINCH 2014.
- 4 Bhalotra, Sonia, Martin Karlsson and Therese Nilsson. Life Expectancy and Mother-Baby Interventions: Evidence from A Historical Trial. CINCH 2014.
- Goebel, Jan, Christian Krekel, Tim Tiefenbach and Nicolas R. Ziebarth.
 Natural Disaster, Environmental Concerns, Well-Being and Policy
 Action: The Case of Fukushima. CINCH 2014.
- 6 Avdic, Daniel, A matter of life and death? Hospital Distance and Quality of Care: Evidence from Emergency Hospital Closures and Myocardial Infarctions. CINCH 2015.
- 7 Costa-Font, Joan, Martin Karlsson and Henning Øien. Informal Care and the Great Recession. CINCH 2015.
- 8 Titus J. Galama and Hans van Kippersluis. A Theory of Education and Health. CINCH 2015.
- **9** Dahmann, Sarah. **How Does Education Improve Cognitive Skills?:** Instructional Time versus Timing of Instruction. CINCH 2015.
- 10 Dahmann, Sarah and Silke Anger. The Impact of Education on Personality: Evidence from a German High School Reform. CINCH 2015.
- **11** Carbone, Jared C. and Snorre Kverndokk. **Individual Investments in Education and Health.** CINCH 2015.
- **12** Zilic, Ivan. **Effect of forced displacement on health.** CINCH 2015.

- De la Mata, Dolores and Carlos Felipe Gaviria. Losing Health Insurance
 When Young: Impacts on Usage of Medical Services and Health. CINCH 2015.
- **14** Tequame, Miron and Nyasha Tirivayi. **Higher education and fertility:** Evidence from a natural experiment in Ethiopia. CINCH 2015.
- **15** Aoki, Yu and Lualhati Santiago. **Fertility, Health and Education of UK Immigrants:** The Role of English Language Skills. CINCH 2015.
- **16** Rawlings, Samantha B., **Parental education and child health:** Evidence from an education reform in China. CINCH 2015.
- 17 Kamhöfer, Daniel A., Hendrik Schmitz and Matthias Westphal.
 Heterogeneity in Marginal Non-monetary Returns to Higher Education. CINCH 2015.
- Ardila Brenøe, Anne and Ramona Molitor. Birth Order and Health of Newborns: What Can We Learn from Danish Registry Data? CINCH 2015.
- **19** Rossi, Pauline. **Strategic Choices in Polygamous Households:** Theory and Evidence from Senegal. CINCH 2016.
- 20 Clarke, Damian and Hanna Mühlrad. The Impact of Abortion Legalization on Fertility and Maternal Mortality: New Evidence from Mexico. CINCH 2016.
- **21** Jones, Lauren E. and Nicolas R. Ziebarth. **US Child Safety Seat Laws:** Are they Effective, and Who Complies? CINCH 2016.
- 22 Koppensteiner, Martin Foureaux and Jesse Matheson. Access to Education and Teenage Pregnancy. CINCH 2016.
- 23 Hofmann, Sarah M. and Andrea M. Mühlenweg. Gatekeeping in German Primary Health Care – Impacts on Coordination of Care, Quality Indicators and Ambulatory Costs. CINCH 2016.
- 24 Sandner, Malte. Effects of Early Childhood Intervention on Fertility and Maternal Employment: Evidence from a Randomized Controlled Trial. CINCH 2016.
- 25 Baird, Matthew, Lindsay Daugherty, and Krishna Kumar. Improving Estimation of Labor Market Disequilibrium through Inclusion of Shortage Indicators. CINCH 2017.
- 26 Bertoni, Marco, Giorgio Brunello and Gianluca Mazzarella. Does postponing minimum retirement age improve healthy behaviors

before retirement? Evidence from middle-aged Italian workers. CINCH 2017.

- 27 Berniell, Inés and Jan Bietenbeck. The Effect of Working Hours on Health. CINCH 2017.
- 28 Cronin, Christopher, Matthew Forsstrom, and Nicholas Papageorge.
 Mental Health, Human Capital and Labor Market Outcomes. CINCH 2017.
- Kamhöfer, Daniel and Matthias Westphal. Fertility Effects of College
 Education: Evidence from the German EducationI Expansion. CINCH 2017.
- **30** Jones, John Bailey and Yue Li. **The Effects of Collecting Income Taxes on Social Security Benefits.** CINCH 2017.
- 31 Hofmann, Sarah and Andrea Mühlenweg. Learning Intensity Effects in Students' Mental and Physical Health – Evidence from a Large Scale Natural Experiment in Germany. CINCH 2017.
- 32 Vollmer, Sebastian and Juditha Wójcik. The Long-term Consequences of the Global 1918 Influenza Pandemic: A Systematic Analysis of 117 IPUMS International Census Data Sets. CINCH 2017.
- **33** Thamarapani, Dhanushka, Rockmore, Marc, and Willa Friedman. **The Educational and Fertility Effects of Sibling Deaths**. CINCH 2018.
- **34** Lemmon, Elizabeth. **Utilisation of personal care services in Scotland:** the influence of unpaid carers. CINCH 2018.
- **35** Avdic, Daniel, Büyükdurmus, Tugba, Moscelli, Giuseppe, Pilny, Adam, and Ieva Sriubaite. **Subjective and objective quality reporting and choice of hospital:** Evidence from maternal care services in Germany. CINCH 2018.
- **36** Hentschker, Corinna and Ansgar Wübker. **Quasi-experimental** evidence on the effectiveness of heart attack treatment in Germany. CINCH 2018.
- Pasha, Mochamad, Rockmore, Marc, and Chih Ming Tan. Early Life
 Exposure to Above Average Rainfall and Adult Mental Health. CINCH 2018
- 38 Elsner, Benjamin and Florian Wozny. The Human Capital Cost of Radiation: Long-run Evidence from Exposure outside the Womb. CINCH 2019
- **39** de la Mata, Dolores and Carlos Felipe Gaviria Garcés. **Exposure to Pollution and Infant Health:** Evidence from Colombia. CINCH 2019
- **40** Besstremyannaya, Galina and Sergei Golovan. **Physicians' altruism in incentives contracts:** Medicare's quality race. CINCH 2019

- 41 Hayen, Arthur P., Klein, Tobias J., and Martin Salm. Does the framing of patient cost-sharing incentives matter? The effects of deductibles vs. no-claim refunds. CINCH 2019
- 42 Molina, Teresa. Pollution, Ability, and Gender-Specific Responses to Shocks. CINCH 2019
- **43** Fischer, Martin, Gerdtham, Ulf-G, Heckley, Gawain, Karlsson, Martin, Kjellsson, Gustav, and Therese Nilsson. **Education and Health: Longrun Effects of Peers, Tracking and Years.** CINCH 2019
- **44** Bannenberg, Norman, Førland, Oddvar, Iversen, Tor, Karlsson, Martin, and Henning Øien. **Preventive Home Visits.** CINCH 2019
- **45** Pestel, Nico and Florian Wozny. **Low Emission Zones for Better Health:** Evidence from German Hospitals. CINCH 2019
- **46** Bartoli, Paola, Grembi, Veronica, and The Linh Bao Nguyen. **Ethnic Density and Health at Birth.** CINCH 2019
- **47** Atal, Juan Pablo, Fang, Hanming, Karlsson, Martin, and Nicolas R. Ziebarth. **Long-Term Health Insurance:** Theory Meets Evidence. CINCH 2020
- 48 Briody, Jonathan. Parental Unemployment During the Great Recession and Childhood Adiposity. CINCH 2020
- **49** Miller, Grant, Paula, Áureo de, and Christine Valente. **Subjective Expectations and Demand for Contraception.** CINCH 2020
- 50 Kronenberg, Christoph. New(spaper) Evidence of a Reduction in
 Suicide Mentions during the 19th-century Gold Rush. CINCH 2020
- 51 Avdic, Daniel, Ivets, Maryna, Lagerqvist, Bo, and Ieva Sriubaite.
 Providers, Peers and Patients: How do Physicians' Practice Environments Affect Patient Outcomes? CINCH 2021
- **52** Alexander Ahammer, Dominik Grübl and Rudolf Winter-Ebmer. **The Health Externalities of Downsizing.** CINCH 2021
- 53 Emilia Barili, Paola Bertoli, Veronica Grembi, and Veronica Rattini.
 COVID Angels Fighting Daily Demons? Mental Health of Healthcare Workers and Religion. CINCH 2021










